From the Department of Women's and Children's Health Karolinska Institutet, Stockholm, Sweden

PRESCHOOL ASTHMA: INFANT LUNG FUNCTION, INFLAMMATION AND SKIN BARRIER FUNCTION IN EARLY LIFE

Martin Färdig



Stockholm, 2023

All previously published papers were reproduced with permission from the publisher. Published by Karolinska Institutet. Printed by Universitetsservice US-AB, 2023 © Martin Färdig, 2023 ISBN 978-91-8017-055-0 Cover illustration: Generated by Martin Färdig with the assistance of DALL·E

Preschool asthma: infant lung function, inflammation and skin barrier function in early life

THESIS FOR DOCTORAL DEGREE (Ph.D.)

By

Martin Färdig

The thesis will be defended in public at Karolinska Institutet in Solna at Rockefeller auditorium, September 19, at 09.00.

Principal Supervisor: Associate professor Björn Nordlund Karolinska Institutet Department of Women's and Children's Health

Co-supervisor(s): Associate Professor Cilla Söderhäll Karolinska Institutet Department of Women's and Children's Health

Professor Karin Cecile Lødrup Carlsen University of Oslo Institute of Clinical Medicine Faculty of Medicine

Professor Catarina Almqvist Karolinska Institutet Department of Medical Epidemiology and Biostatistics *Opponent:* Associate professor Varpu Elenius University of Turku Department of Pediatrics

Examination Board: Associate professor Anna Schandl Karolinska Institutet Department of Molecular Medicine and Surgery

Associate professor Linnea Hedman Umeå University Department of Public Health and Clinical Medicine

Associate professor Apostolos Bossios Karolinska Institutet Department of Respiratory Medicine and Allergy & Department of Medicine

POPULAR SCIENCE SUMMARY OF THE THESIS

Asthma is one of the most common respiratory disorders in children, characterized by airway obstruction leading to symptoms such as wheeze, cough and shortness of breath, negatively affecting daily life of many children. While the cause of asthma remains largely unknown, the disorder likely begins early in life. In this project, we first investigated if early signs of airway obstruction by lower lung function was linked to asthma development at preschool age. Second, we explored if skin-related conditions in early life – such as the skin's ability to hold moist, or eczema with dry, itchy and red skin, or mutations in a gene essential for the skin barrier's normal function – were related to preschool asthma development. Third, we aimed to identify if early-life inflammation – such as higher levels of an inflammatory substance released by white blood cells (eosinophils), commonly increasing during allergic reactions – was connected to asthma development among preschoolers. Since lower lung function (1), skin-related conditions (2) and inflammation by higher levels of the inflammatory substance (3) previously have been associated with asthma among older children, adolescents and adults, the overall purpose was to explore if these three potential risk factors (1-3) in early life was linked to preschool asthma in almost 2400 boys and girls participating in a study in Norway and Sweden, followed from birth to preschool age.

1. With infant lung function measured both while awake and sleeping, we first determined whether lung function differed between the two arousal states. Infant lung function was lower while sleeping compared to the awake state, suggesting it is important to record lung function separately according to arousal state. In the awake state, lower infant lung function was linked to a five times higher likelihood of preschool asthma. However, separating boys and girls, lower infant lung function was only linked to asthma in boys, with an eight times higher likelihood.

2. In regard to skin-related conditions, neither eczema nor mutations in the skin barrier gene were overall associated with preschool asthma. However, in infancy, lower ability of the skin to hold moist was associated with a two times higher likelihood of developing preschool asthma, suggesting a possible link between skin barrier impairment and respiratory health. Interestingly, separating boys and girls, mutations in the skin barrier gene were related to a three times higher likelihood of preschool asthma in girls only.

3. Concerning the inflammatory substance released by white blood cells, higher levels measured in blood in late infancy and at preschool age were connected to preschool asthma, increasing the likelihood by two and almost five times at the two ages, respectively. When separating girls and boys, elevated levels of the same inflammatory substance in infancy increased the likelihood by four times in boys only, whereas at preschool age, elevated levels increased the likelihood in both sexes.

In summary, based on our findings we conclude that lower lung function, skin-related conditions as well as inflammation in early life overall may be linked to preschool asthma

development. However, the likelihood of preschool asthma differed between boys and girls, suggesting sex is an important factor.

ABSTRACT

Background

Asthma is among the most common obstructive respiratory disorders in children, characterized by airway inflammation with features such as wheeze, cough and shortness of breath. Though the etiology behind the disease partly remains unknown, the roots likely origin from early life. Lower lung function, skin barrier impairment, as well as eosinophil inflammation are factors related to asthma development, but are largely unexplored in early life. The overall aim of this doctoral thesis was to explore early risk factors for the development of preschool asthma, related to infant lung function, eosinophil inflammation and skin barrier function in early life.

Methods

All four sub studies (I-IV) were based on data prospectively collected from children actively participating in the Preventing Atopic Dermatitis and ALLergies in children (PreventADALL) mother-child cohort (n=2394) recruited from the general population in Sweden and Norway.

First, with infant lung function measured in both the awake and sleeping state at three months, we aimed to determine whether lung function by tidal flow-volume (TFV) loops differed according to arousal state (**sub study I**). Thereafter, we investigated the potential association between infant lung function while awake and preschool asthma at three years (**sub study IV**).

Second, we aimed to explore if infant skin barrier function by transepidermal water loss (TEWL), eczema at three months and mutations in the skin barrier filaggrin (*FLG*) gene were associated with infant lung function at three months (**sub study II**) and/or preschool asthma at three years (**sub study IV**).

Third, we aimed to identify if higher levels of inflammation by serum eosinophil-derived neurotoxin (EDN) at one and three years were associated with preschool asthma at three years (**sub study III**).

Lastly, we aimed to assess if the potential associations between infant lung- and skin barrier function as well as early-life eosinophil inflammation and preschool asthma differed by sex (**sub study III-IV**).

Results

Among 91 infants with paired lung function measurements, TFV loop parameters overall differed, with a higher ratio of time to peak tidal expiratory flow to expiratory time, t_{PTEF}/t_E , shorter t_E but similar t_{PTEF} while awake compared to the sleeping state. Studying 563 children with information on infant lung function while awake and preschool asthma, a five-

respectively three-fold higher odds ratio (OR) of asthma was observed among infants with lower $t_{\text{PTEF}}/t_{\text{E}}$ (<0.25) and shorter t_{PTEF} (<0.17 s).

While the presence of eczema or FLG mutations were not associated with a shorter t_{PTEF} in 899 infants, a high TEWL was. Similarly, a two-fold higher OR of preschool asthma was observed in infants with high TEWL, but not eczema or FLG mutations among 1337 children.

Among 1233 children with available serum eosinophil-derived neurotoxin (EDN) levels, EDN levels \geq 26.7 µg/L in late infancy and \geq 20.5 µg/L at preschool age were associated with preschool asthma, increasing the OR by two- and almost five-fold at respective age.

Overall, boys had higher rates of preschool asthma compared to girls; $\geq 15\%$ versus $\leq 9\%$. Similarly, infant lung function was lower in boys compared to girls; 0.38 versus 0.40 for t_{PTEF}/t_E and 0.20 versus 0.21 s for t_{PTEF} , whereas TEWL, eczema and *FLG* mutations were distributed equally across the sexes. Boys had higher EDN levels in late infancy and at preschool age (32.0 and 20.9 µg/L) compared to girls (24.5 and 19.0 µg/L). The associations to preschool asthma differed by sex, with an eight- and three-fold higher OR in boys with a lower t_{PTEF}/t_E respectively shorter t_{PTEF} , and a four-fold higher OR observed in girls with *FLG* mutations, only, whereas neither eczema nor high TEWL remained significant risk factors. Higher EDN levels in late infancy increased the OR of preschool asthma by four-fold in boys only, whereas at preschool age, higher levels increased the OR in both sexes.

Conclusion

Based on our findings, we conclude that lower lung function in the awake state, skin barrier impairment as well as eosinophil inflammation in early life overall may be linked to preschool asthma. However, as the associations to preschool asthma differed by sex, with a higher OR observed in boys with lower lung function, while *FLG* gene mutations increased the OR in girls, and in late infancy, higher EDN levels remained a significant risk factor in boys, only, our observations point to the importance of potential sex differences.

LIST OF SCIENTIFIC PAPERS AND MANUSCRIPTS INCLUDED IN THE THESIS

- Bains KES*, Färdig M*, Gudmundsdóttir HK, Almqvist C, Hedlin G, Nordhagen LS, Rehbinder EM, Skjerven HO, Söderhäll C, Vettukattil R, Nordlund B, Lødrup Carlsen KC. Infant tidal flow-volume parameters and arousal state. ERJ Open Res. 2022 Oct 17;8(4):00163-2022. doi: 10.1183/23120541.00163-2022. PMID: 36267897; PMCID: PMC9574559.
 * Shared first authorship
- II. Färdig M, Gudmundsdóttir HK, Hoyer A, Bains KES, Almqvist C, Monceyron Jonassen C, Rehbinder EM, Skjerven HO, Staff AC, Vettukattil R, Söderhäll C, Carlsen KCL, Nordlund B. Skin Barrier Function and Infant Tidal Flow-Volume Loops-A Population-Based Observational Study. Children (Basel). 2022 Dec 31;10(1):88. doi: 10.3390/children10010088. PMID: 36670639; PMCID: PMC9856825.
- III. Färdig M, Lie A, Borres MP, Ekenkrantz T, Granum B, Haugen G, Jonassen CM, Movérare R, Rehbinder EM, Skjerven HO, Staff AC, Vettukattil R, Lødrup Carlsen KC, Söderhäll C, Nordlund B. Eosinophil derived-neurotoxin levels by sex and in relation to preschool asthma a prospective observational study. [under review]
- IV. Färdig M, Hoyer A, Almqvist C, Bains KES, Lødrup Carlsen KC, Gudmundsdóttir HK, Granum B, Haugen G, Hedlin G, Jonassen CM, Konradsen JR, Lie A, Rehbinder EM, Skjerven HO, Staff AC, Vettukattil R, Söderhäll C, Nordlund B. Infant lung function and early skin barrier impairment in the development of preschool asthma. [under review]

Scientific papers not included in the thesis:

- I. Tedner SG, Söderhäll C, Konradsen JR, Bains KES, Borres MP, Carlsen KH, Carlsen KCL, Färdig M, Gerdin SW, Gudmundsdóttir HK, Haugen G, Hedlin G, Jonassen CM, Kreyberg I, Mägi CO, Nordhagen LS, Rehbinder EM, Rudi K, Skjerven HO, Staff AC, Vettukattil R, van Hage M, Nordlund B, Asarnoj A. Extract and molecular-based early infant sensitization and associated factors-A PreventADALL study. Allergy. 2021 Sep;76(9):2730-2739. doi: 10.1111/all.14805. Epub 2021 May 4. PMID: 33751598.
- II. Hoyer A, Rehbinder EM, Färdig M, Asad S, Lødrup Carlsen KC, Endre KMA, Granum B, Haugen G, Hedlin G, Monceyron Jonassen C, Katayama S, Konradsen JR, Landrø L, LeBlanc M, Olsson Mägi CA, Rudi K, Skjerven HO, Staff AC, Vettukattil R, Bradley M, Nordlund B, Söderhäll C. Filaggrin mutations in relation to skin barrier and atopic dermatitis in early infancy. Br J Dermatol. 2022 Mar;186(3):544-552. doi: 10.1111/bjd.20831. Epub 2021 Dec 22. Erratum in: Br J Dermatol. 2022 Jul;187(1):132. PMID: 34698386.
- III. Wärnberg Gerdin S, Lie A, Asarnoj A, Borres MP, Lødrup Carlsen KC, Färdig M, Konradsen JR, Monceyron Jonassen C, Olsson Mägi CA, Rehbinder EM, Rudi K, Skjerven HO, Staff AC, Söderhäll C, Tedner SG, van Hage M, Vettukattil R, Nordlund B. Impaired skin barrier and allergic sensitization in early infancy. Allergy. 2022 May;77(5):1464-1476. doi: 10.1111/all.15170. Epub 2021 Nov 12. PMID: 34738238.
- IV. Skjerven HO, Lie A, Vettukattil R, Rehbinder EM, LeBlanc M, Asarnoj A, Carlsen KH, Despriee ÅW, Färdig M, Gerdin SW, Granum B, Gudmundsdóttir HK, Haugen G, Hedlin G, Håland G, Jonassen CM, Landrø L, Mägi CO, Olsen IC, Rudi K, Saunders CM, Skram MK, Staff AC, Söderhäll C, Tedner SG, Aadalen S, Aaneland H, Nordlund B, Lødrup Carlsen KC. Early food intervention and skin emollients to prevent food allergy in young children (PreventADALL): a factorial, multicentre, clusterrandomised trial. Lancet. 2022 Jun 25;399(10344):2398-2411. doi: 10.1016/S0140-6736(22)00687-0. PMID: 35753340.
- V. Bains KES, Gudmundsdóttir HK, Färdig M, Amnö E, Jonassen CM, Nordlund B, Rehbinder EM, Skjerven HO, Rueegg CS, Vettukattil R, Lødrup Carlsen KC. Infant lung function: criteria for selecting tidal flow-volume loops. ERJ Open Res. 2022 Oct 24;8(4):00165-2022. doi: 10.1183/23120541.00165-2022. PMID: 36299362; PMCID: PMC9589321.
- VI. Gudmundsdóttir HK, Carlsen OCL, Bains KES, Färdig M, Haugen G, Jonassen CM, LeBlanc M, Nordlund B, Rehbinder EM, Skjerven HO, Staff AC, Vettukattil R, Lødrup Carlsen KC. Infant lung function and maternal physical activity in the first half of pregnancy. ERJ Open Res. 2022 Oct 31;8(4):00172-2022. doi: 10.1183/23120541.00172-2022. PMID: 36329799; PMCID: PMC9619250.
- VII. Gudmundsdóttir HK, Hilde K, Bains KES, Färdig M, Haugen G, LeBlanc M, Nordhagen LS, Nordlund B, Rehbinder EM, Skjerven HO, Staff AC, Vettukattil R, Carlsen KCL. Fetal thoracic circumference in mid-pregnancy and infant lung function. Pediatr Pulmonol. 2023 Jan;58(1):35-45. doi: 10.1002/ppul.26153. Epub 2022 Oct 11. PMID: 36097818.
- VIII. Hilde K, Gudmundsdóttir HK, Bains KES, Carlsen KCL, Färdig M, Jonassen CM, Kreyberg I, LeBlanc M, Nordlund B, Rehbinder EM, Roland MCP, Skjerven HO, Staff AC, Vettukattil R, Haugen G. Third trimester fetal lung volume, thoracic circumference, and early infant lung function. Pediatr Pulmonol. 2023 Jul;58(7):2050-2058. doi: 10.1002/ppul.26431. Epub 2023 Apr 26. PMID: 37098847.

IX. Hilde K, Gudmundsdóttir HK, Stensby Bains KE, Färdig M, Lødrup Carlsen KC, Jonassen CM, Kreyberg I, Nordlund B, Rehbinder EM, Paasche Roland MC, Skjerven HO, Staff AC, Vettukattil R, Haugen G. Fetal pulmonary artery Doppler blood flow velocity measures and early infant lung function. A prospective cohort study. J Matern Fetal Neonatal Med. 2023 Dec;36(1):2213796. doi: 10.1080/14767058.2023.2213796. PMID: 37197978

CONTENTS

1	LITERATURE REVIEW 1					
	1.1	Presch	ool asthma	1		
		1.1.1	Diagnosis of preschool asthma	1		
		1.1.2	Phenotypes of wheeze and preschool asthma	1		
		1.1.3	Etiology of asthma and disease development in early life	2		
		1.1.3.1	Infant lung function	. 4		
		1.1.3.2	PInfant skin barrier function	. 7		
		1.1.3.3	B Early-life eosinophil inflammation	. 9		
		1.1.3.4	Sex differences	11		
2	RES	EARCH	I AIMS	13		
3	MAT	MATERIALS AND METHODS				
	3.1	STUD	Y DESIGN	15		
		3.1.1	The PreventADALL study	15		
	3.2	DATA	COLLECTION	17		
		3.2.1	Follow-up and data collection	17		
		3.2.1.1	Infant lung function	18		
		3.2.1.2	Clinical skin assessments	18		
		3.2.1.3	Serum eosinophil-derived neurotoxin and FLG mutations	19		
		3.2.1.4	Preschool asthma	19		
	3.3	3 STUDY POPULATIONS		20		
3.4 Definitions		tions	20			
		3.4.1	Exposures	20		
		3.4.2	Primary outcomes	21		
		3.4.3	Secondary outcomes	21		
	3.5	STAT	ISTICAL ANALYSES	22		
	3.6	Methodological summary		25		
	3.7	ETHICAL CONSIDERATIONS		26		
4	RES	RESULTS				
	4.1	Infant	lung function	27		
		4.1.1	Infant lung function and arousal state	27		
		4.1.2	Infant lung function and preschool asthma	28		
	4.2	Skin b	arrier function	29		
		4.2.1	Skin barrier function and infant lung function	29		
		4.2.2	Infant skin barrier function and preschool asthma	30		
	4.3	Early-l	life eosinophil inflammation	32		
		4.3.1	Early-life eosinophil inflammation and preschool asthma	32		
	4.4	Sex di	fferences	35		
		4.4.1	Infant lung- and skin barrier function, early-life eosinophil			
			inflammation and preschool asthma by sex	35		
5	DISC	CUSSIO		39		
	5.1	Main f	indings	39		

	5.2 Main findings in relation to previous research			
		5.2.1 Infant lung function	41	
		5.2.1.1 Infant lung function and arousal state	41	
		5.2.1.2 Infant lung function and preschool asthma	41	
		5.2.2 Infant skin barrier function	41	
		5.2.2.1 Skin barrier function and infant lung function	41	
		5.2.2.2 Infant skin barrier function and preschool asthma	42	
		5.2.3 Early-life eosinophil inflammation	43	
		5.2.3.1 Early-life eosinophil inflammation and preschool asthma	43	
		5.2.4 Sex differences	43	
		5.2.4.1 Infant lung- and skin barrier function, early-life eosinophil		
		inflammation and preschool asthma by sex	43	
	5.3	Strengths and limitations	44	
6	CON	DNCLUSIONS		
7	POINTS OF PERSPECTIVE		49	
	7.1	Clinical implications	49	
	7.2	Future research	50	
8	ACK	ACKNOWLEDGEMENTS		
9	Financial support			
10	REF	ERENCES	57	

LIST OF ABBREVIATIONS

AD	atopic dermatitis
AS	allergic sensitization
AUC	area under curve
β	beta
CI	confidence interval
DC	dendritic cell
ECP	eosinophil cationic protein
EDN	eosinophil-derived neurotoxin
EPO	eosinophil peroxidase
FLG	filaggrin
GA	gestational age
GDPR	general data protection regulation
g/m²/h	grams per square meter per hour
IgE	immunoglobulin E
IL	interleukin
ILC2	innate lymphoid cells type 2
IQR	interquartile range
kU _A /L	kilounits of allergen specific IgE per liter
MBP	major basic protein
NPV	negative predictive value
OR	odds ratio
PPV	positive predictive value
PreventADALL	Preventing Atopic Dermatitis and ALLergies in children
rBO	recurrent bronchial obstruction
RR	respiratory rate
ROC	receiver operating characteristic
SD	standard deviation
SPT	skin prick test
<i>t</i> E	expiratory time
TEWL	transepidermal water loss

TFV	tidal flow-volume
Th2	t-helper type 2
<i>t</i> _{PTEF}	time to peak tidal expiratory flow
TSLP	thymic stromal lymphopoietin
UKWP	United Kingdom Working Party
VT	tidal volume

1 LITERATURE REVIEW

1.1 PRESCHOOL ASTHMA

Asthma is among the most common chronic obstructive respiratory disorders,¹ characterized by inflammation leading to hyperresponsiveness, reversible airflow obstruction and remodeling of the airway.² The prevalence of asthma among preschool children is around 11%,³ and include common features such as wheeze, cough, chest rigidity and shortness of breath,⁴ of which at least one symptom is present during an asthma attack.⁵ At preschool age, exacerbations are predominately triggered by viral infections⁶ rather than physical activity, allergens, irritants and pollutants,⁴ often observed in older children.⁶ Nonetheless, the morbidity of childhood asthma is high, associated with reduced quality of life, higher school-absenteeism⁷⁻⁹ and hospitalization.^{7,8}

1.1.1 Diagnosis of preschool asthma

Since many children experience recurrent infection-triggered symptoms of wheeze, the most commonly occurring and specific symptom of asthma in young children,⁵ decisions regarding diagnosis and treatment can be difficult as not all wheezing indicates asthma.¹⁰ In contrast to school-aged children, adolescents and adults, lung function testing supporting asthma diagnosis is not routine in infants and preschool children, due to the challenges accomplishing reproducible measurements. Instead, the diagnostic criteria of asthma in preschool children is mainly based on patterns of symptoms,¹⁰ often reported by parents at examination⁵ (such as persistent wheezing or coughing, nocturnal exacerbations, physical or social inactiveness), identified risk factors (such as history or heredity of allergic diseases), response to asthma treatment and exclusion of alternative diagnosis.¹⁰ Although most children develop asthma in early life, many outgrow their asthma-like symptoms at preschool age. Therefore, many clinicians avoid labelling children with "asthma" at this age, while others promote using the term as an umbrella diagnosis gathering children with similar symptoms but with uncertain development.¹¹ Supporting the need for a disease label at preschool age, a recent birth cohort study reported a 20-150-fold increased risk of asthma at five years in children wheezing during first years of life.¹²

Treating asthma-like symptoms in preschool children aims to prevent exacerbations and improve control,¹³ while reducing treatment delay and morbidity burden.¹⁴ The standard treatment of asthma includes inhaled bronchodilators for symptom relief, and anti-inflammatory drugs such as corticosteroids,^{15,16} and in serious cases the addition of immunomodulatory medication, for control.^{17,18}

1.1.2 Phenotypes of wheeze and preschool asthma

Young children have considerably smaller airways compared to older children and adults, which in the presence of inflammation may lead to airway obstruction and wheeze. In many children, wheezing is the first episode of asthma. About one third of all wheezing preschool children have persisting symptoms and develop asthma by school age. Thus, the classification of wheezing phenotypes is important to identify children at risk of asthma.¹⁴ Based on symptoms, time of onset and/or atopic status,¹⁹ there are at least three phenotypes of childhood wheeze; 1) Early transient wheeze; which is manifested in the first three years of life and resolves before six. 2) Persisting wheeze; in which episodes also begin before age three but persists after six. 3) Late-onset wheeze; with symptoms occurring between three and six years of age.⁵ Yet, the stability and clinical usefulness of these phenotypes are debateable.¹⁰ However, both persisting and late-onset wheezers are more likely to have persisting symptoms into adolescence and adulthood,²⁰ with an increasing risk of developing asthma for every year older they get,⁵ while early onset wheezing is linked to comorbidity of other allergic diseases in childhood.⁴

1.1.3 Etiology of asthma and disease development in early life

Childhood asthma is heterogeneous, and the etiology behind the disease is partly still unknown.⁷ Studies suggest that the roots of the disease origin from early life, even though it may become clinically evident first many years later.^{1,7}

First, asthma is a known heritable disease estimated to account for a large part of the disease burden in childhood.^{9,21} Though poorly understood, combinations of genetic susceptibility and environmental factors appear to be important determinants,²² where exposures in prenatal and postnatal life may have lifelong effect on respiratory health^{1,23} in which children are believed to be particularly sensitive, where timing of exposure seems critical.²⁴ Among early-life risk factors maternal nicotine use and diet during pregnancy, cesarean section delivery, preterm birth, the use of antibiotics and antipyretics, exposure to mold, mites, animal dander, and other allergens have been identified.⁶ While some environmental exposures such as tobacco smoke are avoidable, contact to allergens or viruses are more difficult to avert.¹ As lifestyles become more westernized in low-income countries the prevalence of asthma also rapidly increases, a global pattern demonstrating the impact of environmental factors.⁴ Linked to higher risk of childhood asthma among socially disadvantaged children,^{25,26} a lack of socioeconomical and environmental stability may additionally potentiate the effect of environmental exposures (**Figure 1**).²⁷



Figure 1. The potential influence of genetics as well as pre- and postnatal environmental factors on respiratory health in offspring. The original illustration by Svanes et al. has been modified.²³

* The illustration was created with Biorender.com

Second, allergic comorbidity is common^{28,29} and often develops in a time-based order,³⁰ supporting the theory that the conditions are closely connected to each other.^{28,29} This temporal progression is commonly referred to as the *atopic march*, beginning in infancy with manifestations of atopic dermatitis (AD) and allergic sensitization (AS) to food, followed by asthma and allergic rhinitis at preschool- and school age (**Figure 2**).^{30,31} However, the clinical characteristics of allergic diseases often differ between individuals and age.^{32,33} While some allergic conditions resolve with age, others persist throughout childhood into adulthood.³⁰ In a recent review, the authors established that AD, asthma and allergic rhinitis often co-exist in childhood, but found no evidence of the hypothesized sequential development (atopic march) explaining the heterogeneity of allergic diseases, concluding that the relationship between the individual diseases ought to be interpreted from a multimorbidity framework, rather than a progressive or causal one.³⁴

Atopic dermatitis is an allergic inflammatory condition characterized by red, inflamed and itchy skin,^{35,36} with the first signs in infancy often occurring during the second postnatal month.³⁷

Food allergy is defined as an adverse immune response after intake of food leading to a hypersensitivity reaction,³⁸ in Western countries affecting approximately 2.0% in infancy and 8.0% at preschool age,³⁹ in which specific immunoglobulin E (IgE) antibodies to common allergens such as cow's milk, egg, nuts, wheat, soy, fish and shell fish are often the cause.³⁸ By performing skin prick tests or analyzing serum for IgE antibodies to specific allergens, it is possible to detect and identify possible allergies.²

In contrast to AD and food allergies, *allergic rhinitis* often emerges later in childhood, affecting between 5.0-8.5% of all preschool children.⁴⁰ The disease is characterized by inflammation in the nasal mucosa due to allergic sensitization to airborne allergens such as pollens, mold and animal dander, with symptoms including sneezing, itching, runny and stuffy nose.⁴¹



Figure 2. Illustration of the distribution of allergic diseases in the hypothesized atopic march, starting with atopic dermatitis and food allergy in infancy, followed by asthma and allergic rhinitis later in childhood.⁴²

mo: months

* Reproduced with permission from Allergy Asthma Immunol Res. 2019 Jan; 11(1): 4-15. doi: 10.4168/aair.2019.11.1.4

1.1.3.1 Infant lung function

Lower lung function in early life is a potential trajectory for future lung function⁴³ and respiratory disease,^{7,44} and increases the overall mortality and morbidity risk.⁴⁵ The fetal development of the lung starts at five weeks gestational age (GA) and involves five different phases: the embryonic (<6 weeks GA), the pseudoglandular (6-16 weeks GA), the canicular

(16-24 weeks GA), the saccular (24-40 weeks GA) and the alveorization (mostly after birth). In the last phase, alveorization commences and continues to develop postnatally the first two to four years in life,¹ potentially throughout the entire childhood into adulthood.^{1,44} Lung growth is rapid in childhood, developing proportionately after body size influenced by age, sex and ethnicity,⁴⁶ resulting in a 20-fold increase of lung function in early adolescence.⁴⁷ Lung function measured in volume reaches a peak in late adolescence and early adulthood, a plateau lasting for many years, before it progressively begins to decrease⁴³ due to lung elasticity loss.⁴⁸ This pattern of growth and decline is a determinant of future lung function.⁴³

The lung's capacity for repair can be negatively affected by peri- and postnatal insults,^{1,44} such as perinatal adverse events, environmental exposures and airway infections,⁴⁴ leaving the child more vulnerable to later respiratory disease. Airway epithelial cell damage is critical in lung development, potentially causing inflammation and cell death. If the subsequent cell regeneration is unsuccessful, remodeling with abnormal tissue structures may follow, characteristic for several respiratory diseases.¹ Children born with low lung function often retain a low lung function later in life. Duration, type and severity of insult influence the timing and outcome of abnormal lung development.⁴⁶

Existing guidelines advise tailoring asthma treatment first handedly after asthmatic symptoms, and to a smaller extent, lung function. Nevertheless, objective lung function measurements are an important tool in diagnosing, monitoring and optimizing therapy of asthmatic children. Controversially, only a fraction of asthmatic children is offered lung function measurements.⁴⁹

Reference values for lung function measurements are obtained from healthy children, aiming to present normal lung function, where body size and age are used as proxies for chest size and maturity.⁴⁸ With reliable reference values largely lacking⁴⁸ and objective lung function measurements being difficult to perform on young children, potential diagnoses are often based on symptoms reported by parents.⁷ While it is possible to obtain lung function tests from older children actively participating, it has historically been challenging assessing lung function in infants and young children without sedation. Commonly, chloral hydrate has been used as a sedative⁵⁰ in in- and outpatient facilities;⁵¹ a drug negatively affecting normal ventilation,⁵² associated with several cases of overdosing, respiratory depression, cardiopulmonary arrest and fatal events.⁵¹ Studies suggest future research ought to concentrate on lung function techniques, allowing repeated measurements in awake young children.⁴⁹

Assessing lung function in awake young children, tidal breath flow-volume (TFV) loops can be analyzed ^{49,53}, presenting compound measurements including size of airways, mechanical characteristics⁵⁴ and respiratory control⁵⁵ and may expose abnormal patterns of breathing and airway obstruction.⁵⁶ The technique allows lung function measurements regardless of age and ability to cooperate, providing important information in both research and clinical settings. To reduce intra-individual variation, measurements should be performed under same circumstances on each occasion; awake children in a friendly setting⁵³ calmly breathing into a face mask (**Figure 3**).⁵⁷



Figure 3. Tidal flow-volume loop measurement in the awake state at three months.

* Reproduced with parental consent.

With a lower ratio of time to peak tidal expiratory flow (t_{PTEF}) to total expiratory time (t_E), t_{PTEF}/t_E (**Figure 4**), previous studies using TFV loops in the awake state were able to identify infants with lower lung function^{58,59} and older infants with airway obstruction,⁶⁰ distinguish asthmatic from non-asthmatic children,⁶¹ as well as track lung function from birth to adolescence in asthmatic children with allergic comorbidities.⁶² Lower t_{PTEF}/t_E is further associated with preschool wheeze^{63,64} and asthma later in childhood.^{59,65} In addition, infants and children with obstructive respiratory diseases generally reach t_{PTEF} earlier (and

consequently after a smaller expiration volume) during the expiratory phase.⁶⁶ Representing one of the two TFV loop indices comprising the more common $t_{\text{PTEF}}/t_{\text{E}}$ ratio, lower t_{PTEF} values have been reported in infants and children with airway obstruction and asthma.66-68 However, since few longitudinal studies yet have confirmed the usefulness of the method, evidence for routine monitoring with TFV loops is still lacking.⁵⁶ Moreover. existing reference values are largely based upon measurements performed in sleeping or sedated infants.⁶⁹ As merely one study from the 1990's compared TFV loops according to arousal state suggesting separate reference values for sleeping and awake infants may be required,⁷⁰ further research is needed.



Figure 4. Illustration of tidal flow-volume (TFV) loops, including the TFV loop parameters of time to peak tidal expiratory flow (t_{PTEF}) and expiratory time (t_E) comprising the t_{PTEF}/t_E ratio.

Knowledge gap

 \rightarrow Does lung function by TFV loops differ while awake compared to the sleeping arousal state in early infancy, and is lower infant lung function in the awake state associated with the development of preschool asthma?

1.1.3.2 Infant skin barrier function

Upholding the homeostasis by balancing body temperature, hydration, endocrine and immune system,⁷¹ the skin and airway epithelial constitute a natural protective barrier from allergens, microbes and environmental pollutants.⁷² However, modern-day lifestyle factors such as frequent bathing, soap use, heating, air conditioners and dehumidifiers can negatively affect the epithelial barrier function.⁷³ Advances in research suggest skin barrier impairment may contribute to the development of inflammatory diseases, including allergic conditions of the skin and the lungs.^{74,75}

Of all children, about one fifth suffer from AD.^{35,36} In the development of AD, reduced epidermal barrier function and dysregulation of the immune system are important factors, on

which both genetic and environmental factors have influence.⁷⁶ Among genes associated with allergic diseases, Filaggrin (FLG) gene mutations are believed to contribute to the atopic march,²⁸ foremost linked to AD, but also AS⁷⁷ and food allergy, asthma and allergic rhinitis.⁷³ Filaggrin is a protein essential for the skin barrier function by creating a permeable barrier, producing natural emollients protecting from transepidermal water loss (TEWL) and maintaining the pH balance protecting the skin from harmful microbes.^{71,73} FLG mutations increase the TEWL, change the structure of the epidermis and leave the skin dry.⁷¹ Not only does elevated TEWL reflect a dysfunctional skin barrier function, as seen in AD, elevated TEWL can also precede clinical manifestations of the disease.⁷⁸ Since FLG is not expressed in the upper airways,⁷³ though this evidence seems conflicting,⁷⁹ it is likely that innate cells of the skin⁸⁰ and lungs⁸¹ initiate a cascade of complex immunological responses triggered by allergens entered through FLG-deficient skin, leading to AS to specific allergens (IgE), involved in both AD and asthma development (Figure 5).^{30,73} When encountering the same allergens later in life, perhaps through the airways, the body reacts and triggers a local inflammatory process in the lungs resulting in airway obstruction and asthma.^{35,75} In support of this theory, studies have shown that FLG mutations are associated with AD, AS and asthma in children with previous history of AD.82 As the first defense against allergens and other environmental irritants, reduced skin barrier may simply function as a facilitator for AD, leading the pathway from inflammation and sensitization to allergens via the skin to allergic expressions in the respiratory system.^{35,71,83} However, the relationship between high TEWL, eczema and FLG mutations in early life and infant lung function as well as preschool asthma is largely unexplored.

Knowledge gap

 \rightarrow Is skin barrier function by TEWL, eczema and *FLG* mutations in early infancy associated with infant lung function by TFV loops in the awake state, as well as the development of preschool asthma?



Figure 5. Model of the possible contribution of skin barrier impairment through the type II (Th2) inflammation, a complex immune response characterized by the manifestation of Th2 cells and IgE,² present in most individuals suffering from asthma and other allergic diseases.⁸⁴ In the first step, food and inhalant allergens penetrate the skin due to barrier impairment caused by factors such as loss-of-function mutations (R501X, 2282del4, R2447X) in the FLG gene, high TEWL and/or the presence of eczema. Secondly, innate epithelial cells in the skin secrete cytokines (for instance TSLP, interleukin (IL)-25, and IL-33), stimulated by allergens. Further, innate immune cells (such as eosinophils, DC, ILC2, basophils and mast cells) are triggered, releasing cytokines (such as the IL-4) and create Th2 cells producing IgE in the lymphatic organs. The Th2 cells additionally release cytokines (IL-4) to stimulate more eosinophils and ILC2, whereas IgE have the ability to incite basophils and mast cells. Thirdly, this positive feedback loop of IgE to mast cells and basophils causes the phenotype of AD. TSLP, IL-25, IL-33, Th2 cells and IgE are thought to enter the lungs via the blood circulation and hereby lead to the development of asthma. The original figure by Yang et al. has been adapted.³⁰

TEWL: transepidermal water loss, FLG: filaggrin, TSLP: thymic stromal lymphopoietin, IL-33: interleukin-33, IL-25: interleukin-25, IL-4: interleukin-4, DC: dendritic cells, ILC2: innate lymphoid cells type 2 Th2: T-helper type 2 cells, IgE: immunoglobulin E, AD: atopic dermatitis, AS: allergic sensitization * The illustration was created with Biorender.com

1.1.3.3 Early-life eosinophil inflammation

An inappropriate immune response to allergens is the shared characteristics of allergy, and may contribute to the development of allergic diseases, including asthma.⁷⁵ IgE reactions are common, but hypersensitivity responses are not seen in all with allergic diseases.³³ Though

childhood asthma is heterogeneous,⁷ many children with asthma share immunological clinical features causing inflammation, hyperresponsiveness and obstruction of the airway,⁸⁵ in which eosinophils may play an important part.^{86,87}

Eosinophils are a type of white blood cells of the innate immunity with a homeostatic function in the body's immune responses against parasites, bacteria and viruses, but commonly also play a pathologic role in inflammatory diseases, such as asthma.⁸⁸ Allergic eosinophil asthma, characterized by the type II inflammation in the airways, is accountable for approximately half of all asthma cases, and even higher in pediatric populations.⁸⁹ Stimulated by allergens and infections encountered in the epithelial mucosa,⁹⁰ activated eosinophils secrete pro-inflammatory degranulation products, such as the eosinophil-derived neurotoxin (EDN), eosinophil cationic protein (ECP), eosinophil peroxidase (EPO), and major basic protein (MBP), and are found in various body fluids, including nasal mucus, sputum, urine and blood.⁹¹ These degranulation products hold cytotoxic and anti-infectious properties,⁹² which in the lungs cause tissue damage and dysfunction, mucus hypersecretion, airway remodeling⁹³ and inflammation, characteristics also observed in children with asthma.⁸⁶

During the past decades, ECP has previously been of particular interest as a promising biomarker in childhood asthma, today widely characterized and studied.⁹⁴ In preschool- and school children, higher ECP levels have been found in children with wheeze,⁹⁵ asthma,⁹⁶ allergic asthma⁹⁷ as well as AS,⁹⁷ and AD,⁹⁸ associated with wheeze, asthma,^{96,99} asthma severity,^{97,100} as well as AS¹⁰¹, AD severity,¹⁰² and change in lung function by tidal flow-volume (TFV) loops after bronchodilator treatment in wheezing children.¹⁰³

Unlike ECP, serum EDN is less sticky showing negligible tube binding after collection, and may thus require smaller specimen amounts, a beneficial characteristic when sampling young children.⁹³ Furthermore, EDN lacks a clear biological rhythm, and appears to retain stable serum concentrations over time and at different temperatures.¹⁰⁴ Interestingly, higher EDN levels are detectable in tissues and body fluids long before asthmatic symptoms are evident.^{86,90} Compared to healthy preschool- and school children, higher EDN levels have been reported in children with asthma,^{101,105-111} linked to bronchial hyper-responsiveness,^{107,108} higher symptom burden,^{106,107,112} being without corticosteroid treatment,¹⁰¹ as well as AS¹¹³ and AD.¹⁰¹ Similarly, higher EDN levels were found in wheezing children compared to asymptomatic children recuperating after respiratory tract infections¹¹⁴ and related to the number of future episodes of wheeze.¹¹⁵ Thus, EDN may have potential as a biomarker.⁸⁶ However, with reference values lacking,⁹³ and the relationship between early-life EDN levels and the risk of asthma in the general pediatric population remaining undetermined, its role in pediatric clinical practice is unknown.

Knowledge gap

 \rightarrow Is early-life eosinophil inflammation by serum EDN levels associated with the development of preschool asthma?

1.1.3.4 Sex differences

The prevalence of wheeze and asthma is higher in young boys compared to girls.^{4,20,116} One leading theory for this trend is the anatomical differences observed between the sexes. Already in early life, female fetuses to a larger extent display signs of lung maturation, such as mouth movements, in fetal life related to breathing and surfactant production.¹¹⁷ Postnatally, growth of airways is proportional to lung size in girls,¹¹⁷ whereas boys have smaller airways in relation to lung size.⁴ Though likely multifactorial, this structural discrepancy may contribute to the higher asthma prevalence in young boys.⁴ In adolescence this pattern reverses to a female predominance.^{4,116} Although the underlying mechanism for this change in puberty remains undetermined,¹¹⁶ the focus shifts to female sex hormones in combination with sex-specific differences in environmental exposures altering immune system responses.^{20,116,118} Other possible explanations are genetic differences together with clinical characteristics and symptom severity varying by sex and with age, influencing diagnosis and treatment, ultimately leading to the discrepancy observed in reported asthma prevalence.¹¹⁶

Knowledge gap

 \rightarrow Do the possible associations between infant lung function and skin barrier function as well as early-life eosinophil inflammation and the development of preschool asthma differ by sex?

2 RESEARCH AIMS

The overall aim of this doctoral thesis was to explore early risk factors for the development of preschool asthma, related to infant lung function, eosinophil inflammation and skin barrier function in early life (**Figure 6**).

Specific aims		
• To determine if infant lung function differed while awake compared to the sleeping state, and investigate the association between lung function in early infancy and the development of preschool asthma.	I, IV	
• To explore the possible association between skin barrier function and lung function in early infancy as well as the development of preschool asthma.	II, IV	
• To identify if early-life eosinophil inflammation is associated with the development of preschool asthma.	III	
• To assess if the potential associations between infant lung-, and skin barrier function as well as early-life eosinophil inflammation and the development of preschool asthma differed by sex.	III, IV	



Figure 6. Illustration of the sub studies (I-IV) included in the thesis.

* The illustration was created with Biorender.com

3 MATERIALS AND METHODS

3.1 STUDY DESIGN

The four sub studies (I-IV) included in this thesis were observational studies, including data prospectively collected in the Preventing Atopic Dermatitis and ALLergies in children (PreventADALL) birth cohort study.¹¹⁹

3.1.1 The PreventADALL study

The PreventADALL study is a Scandinavian, multicenter, population-based, prospective birth cohort study with two 2x2 factorially designed randomized controlled interventions; skin care and early food introduction, aiming to determine if primary prevention of allergic diseases is possible.¹¹⁹

All pregnant women attending the routine ultrasound screening at approximately 18 weeks GA in Norway (Oslo University Hospital and Østfold Hospital Trust) and Sweden (Karolinska University Hospital, Ultragyn AB, Ultraljudsbarnmorskorna, Solna barnmorskemottagning, Danderyds sjukhus, BB-Sophia, Mama Mia, Södersjukhuset, Södertälje sjukhus and BB-Stockholm) were invited to participate. Exclusion criteria were insufficient language skills (Norwegian or Swedish), planning to move away from study site within the first year postnatally, or pregnancy with more than two fetuses. From December 2014 to October 2016, 2697 women were recruited; 2149 in Norway and 552 in Sweden.¹¹⁹

The last infant was born in April 2017, rendering 2397 mother-child pairs, of which three later withdrew consent. Exclusion criteria were premature birth (<35 weeks GA) and severe malformations or disease. At birth, infants were randomly allocated (1:1:1:1) into one of four groups; "skin intervention" (oil-baths and facial cream at least four days per week, from two weeks to nine months), "food intervention" (complementary feeding of peanut, cow's milk, wheat and egg, from three months), "combined skin and food intervention" or "no intervention" (**Figure 7**).¹¹⁹ Neither the skin, nor the food intervention reduced the risk of AD in the first year of life.¹²⁰ However, early complementary food introduction reduced the risk of soft food allergy at preschool age.¹²¹



Figure 7. *Flowchart of the PreventADALL birth cohort study, illustrating the skin and food interventions, time points for data collection and outcomes.*

3.2 DATA COLLECTION

The four sub studies (I-IV) in this thesis focused on information gathered about the participating PreventADALL children, from early infancy to preschool age (Figure 8).



Figure 8. Timeline of the PreventADALL birth cohort study, including time points for data collection through clinical follow-ups and electronic questionnaires. The four sub studies (I-IV) in the current thesis explored the associations between infant lung function, TEWL, eczema at three months, FLG mutations, as well as EDN levels at one and three years and preschool asthma at three years.

GA: gestational age, TEWL: transepidermal water loss, EDN: eosinophil-derived neurotoxin * The illustration was created with Biorender.com

3.2.1 Follow-up and data collection

Shortly after enrollment, after informed consent was collected, mothers answered two electronic questionnaires at approximately 18 and 34 GA weeks, gathering information on maternal health, including allergic diseases, together with lifestyle and environmental factors and information about the father.

Within the first day after birth, after consent from the parent(s) was given, the infants were included and randomly allocated to one of the PreventADALL intervention groups by study personnel visiting the maternity wards. At the same occasion, birth data, anthropometrics, and biological samples were collected.

Comprehensive electronic questionnaires were sent every third month the first year starting from three months, thereafter twice annually. Clinical follow ups were performed at three, six and 12, 24 and 36 months. Data collection included measurements of weight, length, lung function, skin barrier function, allergic diseases, and biological sampling, all conducted by

trained study personnel following standard operating procedures and the PreventADALL study protocol.

3.2.1.1 Infant lung function

Lung function by TFV loops was measured at the three-, 12- and 36-month clinical followups. In infancy, lung function measurements were overall performed prior to other investigations, in the arousal state of which the infants arrived at the study site to ensure infants were calm. If the first TFV measurement was conducted while the infant was asleep, a second test was attempted in the awake state. Infants were positioned in a supine position in the caregiver's lap or the cot, whichever was successful. Using the employed Exhalyzer® D (Eco Medics AG, Duernten, Switzerland), TFV loops were sampled through an ultrasound flow head and a dead-space reducer, with a carefully fitted face mask with an air-inflated rim, by a stable grip over the mouth and nose of the infant, using fingers to control for minimal leakage from the mask, according to existing guidelines.^{57,122} When the infant breathed

calmly and evenly into the mask, loop sampling started, aspiring to record at least ten consecutive breaths.

All TFV loop measurements were manually analyzed by three reviewers in Oslo (Norway) and Stockholm (Sweden), focusing on loop shape and reproducibility (Figure 9). Technically unacceptable loops were manually removed before approval. Further information about the standard operating procedure for TFV loop measurements and manual loop selection is described elsewhere.¹²³ At three months, 1183 infants had at least one acceptable TFV loop measurement; 899 in the awake state, 375 in the sleeping state, and 91 in both the awake and sleeping state.



Figure 9. Visual description of tidal flow-volume (TFV) loops before and after the manual loop selection.

tPTEF/tE: time to peak tidal expiratory flow to expiratory time

3.2.1.2 Clinical skin assessments

Clinical skin examinations were carried out by trained study personnel at three, six, 12, 24 and 36 months educated together at workshops to minimize inter-observer variability. In infancy, the parents were advised not to bathe their children in oils or apply skin emollients 24 hours prior to the clinical follow-ups.

Clinically observed eczematous skin lesions were confirmed by physicians excluding common differential diagnosis to AD, such as seborrheic and contact dermatitis.¹²⁴ AD was assessed using the United Kingdom Working Party (UKWP) criteria in the first year of life, UKWP and Hanifin and Rajka criteria at the subsequent clinical follow-ups.¹²⁵

TEWL (g/m²/h) was measured at three, six, 12 and 36 months on the left lateral upper arm with an open chamber DermaLab USB (Cortex, Hadsund, Denmark), at room temperature between 20-25 °C and within acceptable ranges of humidity, in line with previous findings.¹²⁶ After 15 minutes of acclimatization, with the children wearing only diapers or underpants, three consecutive measurements were performed in calm children, distanced from direct sunlight, while windows and doors were kept closed.

3.2.1.3 Serum eosinophil-derived neurotoxin and FLG mutations

Blood samples were collected from cord blood at birth and at clinical follow-ups at three, six, 12, 24 and 36 months. All samples were kept in room temperature for 60-90 minutes before centrifugation and separation of serum and plasma, thereafter, kept at -20 °C for five to seven days before being transferred for storage at -80 °C.

Serum EDN at 12 and 36 months was measured using an automated ImmunoCAP EDN research assay (Thermo Fisher Scientific, Uppsala, Sweden). Description and performance of the method is reported elsewhere.¹⁰⁹

DNA was extracted from blood and genotyped for the most common European loss-offunction mutations in the *FLG* gene (R501X, 2282del4 and R2447X), using the TaqManbased allelic discrimination assay, as described by Hoyer *et al.*¹²⁷

3.2.1.4 Preschool asthma

Information on doctor-diagnosed asthma and wheeze was collected from questionnaires at three, six, nine, 12, 18, 24, 30 and 36 months and clinical follow-ups at 24 and 36 months. Asthma medication use was retrieved from questionnaires at 12, 18, 24, 30 and 36 months. To compensate for missing information on medication use in some participants, the children's medical records were reviewed for the prescription of asthma medication between nine months through three years.

Allergic sensitization

Skin prick tests (SPT) were performed at three years using standard allergen solutions for egg, cow's milk, peanut, wheat, soy, cod, birch, grass, dog, cat, and house dust mite (Soluprick ALK-Albelló, Hørsholm, Denmark). In addition, at the same time point analyses for specific serum immunoglobulin E (IgE) to egg, cow's milk, peanut, wheat, soy, cod, birch, grass, mugwort, dog, cat, horse, house dust mite, and mold (Cladosporium herbarium) were performed using the ImmunoCAP (Thermo Fisher Scientific).

3.3 STUDY POPULATIONS

The study populations of the included sub studies (I-IV) are shown in Figure 10.

Sub study I: 91 infants with available infant lung function measurements (TFV loops) in the awake and sleeping state at three months.

Sub study II: 899 infants with available infant lung function measurements (TFV loops) in the awake state as well as either TEWL, eczema at three months and/or *FLG* mutations.

Sub study III: 1233 children with available serum EDN levels at one and/or three years.

Sub study IV: 1337 children with available information related to the presence or absence of preschool asthma at three years, as well as either infant lung function measurements (TFV loops) in the awake state, TEWL, eczema at three months and/or *FLG* mutations.



Figure 10. Flow chart of the study populations in the four sub studies (I-IV) included in this thesis.

TEWL: transepidermal water loss, FLG: filaggrin, EDN: eosinophil-derived neurotoxin

3.4 DEFINITIONS

3.4.1 Exposures

Lower infant lung function

Lower lung function at three months was defined as a $t_{\text{PTEF}/t_{\text{E}}} < 0.25$, previously associated with airway obstruction and asthma.^{59,128-130} In addition, a $t_{\text{PTEF}} < 0.17$ s (<25th percentile) was chosen as secondary exposure, as lower values have previously been found in infants and children with airway obstruction and asthma (**sub study IV**).⁶⁶⁻⁶⁸
High transepidermal water loss

A mean TEWL value >75th percentile denoted an impaired skin barrier function, >8.83 g/m²/h in **sub study II** and >9.50 g/m²/h in **sub study IV** at three months, in line with previous reports.^{126,131}

Eczema

As few three-month-old infants fulfil the strict diagnostic criteria for AD by the United Kingdom Working Party (UKWP),¹³² eczema was used as a proxy. Eczema at three months was defined as having clinically observed eczematous skin lesions with the exclusion of common differential diagnosis to AD, such as seborrheic and contact dermatitis, confirmed by physicians (**sub study II and IV**).¹²⁴

Filaggrin mutations

Filaggrin mutations were defined as being carriers of any of the three mutations (R501X, 2282del4 and R2447X),¹²⁷ hypothesized to contribute to asthma (sub study II and IV).³⁶

Higher eosinophil-derived neurotoxin levels

Higher EDN levels at one and three years equal to EDN levels \geq 26.7 and \geq 20.5 µg/L, respectively (**sub study III**).

3.4.2 Primary outcomes

Lower infant lung function

Lower lung function at three months was defined as a $t_{\text{PTEF}/tE} < 0.25$, previously associated with airway obstruction and asthma (**sub study I and II**).^{59,128-130} In addition, a $t_{\text{PTEF}} < 0.17$ s (<25th percentile) was chosen as secondary outcome, as lower values have previously been found in infants and children with airway obstruction and asthma (**sub study II**).⁶⁶⁻⁶⁸

Preschool asthma

Preschool asthma at three years was defined as having recurrent bronchial obstruction (≥3 episodes of recurrent bronchial obstructions) (rBO) between two and three years, and fulfilling at least one of the following criteria; 1) doctor-diagnosed asthma between zero and three years, and 2) any use of and/or prescribed asthma medications (bronchodilators, inhaled corticosteroids or leukotriene-antagonist) between nine months through three years (**sub studies III and IV**).¹³³

3.4.3 Secondary outcomes

Allergic sensitization

Allergic sensitization at three years was defined as having positive SPT (\geq 3 mm) and/or specific IgE level (>0.35 kU_A/L) towards any allergen (egg, cow's milk, peanut, wheat, soy,

cod, birch, grass, mugwort, dog, cat, horse, house dust mite and/or mold) at three years (sub study III).

Atopic dermatitis

Atopic dermatitis at three years was defined as fulfilling the United Kingdom Working Party and/or Hanifin and Rajka criteria at three years (**sub study III**).¹²⁵

Children with any allergic disease

Any allergic disease at three years was defined as children with either any wheeze, preschool asthma, AS and/or AD at three years (**sub study III**).

Non-atopic children

Non-atopic children at three years were defined as children with no history of any wheeze, doctor-diagnosed asthma, use of and/or prescribed asthma medications, AS or AD between birth and three years (**sub study III**).

3.5 STATISTICAL ANALYSES

Statistical analyses in the four sub studies (I-IV), conducted using IBM SPSS Statistics (version 26 or later) or RStudio (4.0.3), are presented in **Table 1**. Statistical significance level was set to 0.05.

Descriptive

Overall, continuous variables with normal distribution were presented with means and standard deviations (SD) and/or minimum-maximum (min-max), while numbers (n) and percentages (%) were used to describe dichotomous/categorical variables (**sub study I-IV**). In addition, median, interquartile range (IQR) and/or the 95th percentile were provided to describe continuous variables when appropriate (**sub study II**), as well as when presenting data with skewed distribution (**sub study III**).

Continuous variables were compared between independent groups using parametric and nonparametric tests; independent t-tests (parametric) for variables with normal distribution (**sub study II-IV**), while Mann Whitney-U test (non-parametric) was used for data with skewed distributions (**sub study III**). Paired samples of continuous variables with skewed distributions were compared using the non-parametric Wilcoxon's signed-rank test (**sub study I, III**).

The non-parametric Chi² tests or Fisher's exact tests (small sample sizes, expected counts <5) were used to compare distributions of dichotomous/categorical variables between independent groups (**sub study II-IV**). Paired samples of dichotomous variables with skewed distributions were compared using the non-parametric McNemar test (**sub study I**).

Associations

Univariate multivariate logistic regression was used to assess associations to dichotomous outcomes (**sub study II-IV**), while multinominal logistic regression was applied to investigate outcomes with more than two categories (**sub study IV**); all models presented with odds ratios (OR) with 95% confidence intervals (95% CI). In **sub study IV**, the relationship between dichotomized exposures and outcomes was assessed with Spearman's rank correlation, presented with Spearman's Rho coefficient (95% CI). Similarly, in **sub study III**, the relationship between exposure (repeated measurements) and outcome was explored using mixed effects logistic regression, presented with OR (95% CI). Univariate conditional logistic regression was used in **sub study I** to estimate OR (95% CI) for dichotomous outcomes of paired samples.

To study the robustness of the findings (**sub study II**), continuous outcomes were used in univariate and multivariate linear regression models, while the univariate relationship between skewed distributions of continuous exposures at two ages were explored in robust linear regression (**sub study III**), both presented with β coefficients (95% CI).

Primarily based on the literature, potential covariates and confounders to exposures and outcomes were selected in prior to the analyses. However, associations between birth and background characteristics and outcomes were explored to reveal possible effect modifiers, and were included in multivariate regression models when applicable.

Interactions

Possible modification of the skin and/or food intervention on the association between exposures and outcomes were assessed in interaction analyses in **sub studies II-IV**, by including interaction terms in univariate and multivariate logistic regression models.

Predictions

To objectively define cut-off levels with highest sensitivity and specificity of the outcomes, Youden's Index was calculated,¹³⁴ whereafter continuous exposures accordingly were dichotomized (**sub study III**). The predictive accuracy of exposures to differentiate children with the outcomes was tested using receiver operating characteristics (ROC) analyses (**sub study III-IV**), presented with area under the ROC curve (ROC-AUC) with 95% CI, sensitivity, specificity, positive predictive value (PPV) and negative predictive value (NPV) in percentages. PPV and NPV were calculated using Chi² tests.

	S	Sub study I-IV		[V
Characteristics	Ι	II	İII	IV
Descriptive				
Continuous variables				
Independent t-test (independent, parametric)		Х	Х	Х
Mann-Whitney U-test (independent, non-parametric)			Х	
Wilcoxon's signed-rank test (paired, non-parametric)	Х		Х	
Dichotomous/categorical variables				
Chi ² test or Fisher's Exact test (independent, non-parametric)		Х	Х	Х
McNemar test (paired, non-parametric)	Х			
Associations				
Dichotomous/categorical outcomes				
Logistic regression		Х	Х	Х
Multinominal logistic regression				Х
Spearman's rank correlation				Х
Conditional logistic regression (paired)	Х			
Mixed effects logistic regression (repeated measurements)			Х	
Continuous outcomes				
Linear regression (normal distribution)		Х		
Robust linear regression (skewed distribution)			Х	
Interactions				
Logistic regression (interaction terms)		Х	Х	Х
Predictions				
Youden's Index			Х	
ROC analyses			Х	Х

Table 1. Overview of the statistical methods used in the four sub studies (I-IV).

3.6 METHODOLOGICAL SUMMARY

	Sub study I-IV			
Characteristics	Ι	II	III	IV
Study design	Observational study	Observational study	Observational study	Observational study
Cohort	PreventADALL	PreventADALL	PreventADALL	PreventADALL
Study population	n=91	n=899	n=1233	n=1337
Exposure(s)	Arousal state	TEWL, eczema <i>FLG</i> mutations	EDN	Lung function, TEWL, eczema, FLG mutations
Outcome	Lung function	Lung function	Preschool asthma	Preschool asthma
Data source(s)	3-month clinical follow- up	3-month clinical follow- up	12-, 24-, 36- month clinical follow-up	3- 12-, 24-, 36- month clinical follow-up
			Electronic questionnaires 3-36 months	Electronic questionnaires 3-36 months
			Medical records 9 months through 3 years	Medical records 9 months through 3 years
Data analyses	Descriptive	Descriptive	Descriptive	Descriptive
	Associations	Associations	Associations	Associations
		Interactions	Interactions	Interactions
			Predictions	Predictions

Table 2. Methodological summary of the four sub studies (I-IV).

3.7 ETHICAL CONSIDERATIONS

All four sub studies in this thesis were conducted in line with the ethical principles of the Declaration of Helsinki,¹³⁵ and was approved by the Regional Committee for Medical and Health Research Ethics in Norway (2014/518), and the Swedish Ethical Review Authority (2014/2242-31/4) with multiple amendments, including the approval of the clinical follow-up at three years (2018/1437-32). To respect the participants autonomy and right to make informed decisions, written consent was collected from all mothers at enrolment, and from parent(s) at infant inclusion, which for the children at present acts as a surrogate consent. Acknowledging children as a vulnerable group, weighing risks and benefits, the trained investigators always discussed painful or discomfortable procedures with parents and tried to prepare the children in advance when possible. For instance, to minimize pain and discomfort a local aesthetic cream was always applied prior to venipuncture, numbing the skin. Aiming to make children feel calm and safe at the time for blood sampling, they were placed in the laps of the parents, often eating a fruit or a biscuit, watching children's television programs. The procedure is relatively quick, and only one attempt was allowed if the children were uneasy or the possibilities to sample blood were considered low. Children unsettled or unwilling to perform certain parts included in the clinical follow-up were never forced to complete the entire investigation. Ongoingly, parents are informed about the progress and novel research results published within the PreventADALL study.

To ensure the security of the study participants data, in compliance with the General Data Protection Regulation (GDPR) legislation, the retrieved information is pseudonymized and located on a secure remote desktop, only accessible for students and supervisors within the PreventADALL research group.

4 RESULTS

Main findings of the four included sub studies (I-IV):

4.1 INFANT LUNG FUNCTION

4.1.1 Infant lung function and arousal state

To determine if lung function differed by arousal state in early infancy, we compared TFV loops while awake to the sleeping state in 91 healthy infants at three months (**sub study I**).

In the 91 infants with TFV loops in both arousal states, t_{PTEF}/t_E was higher, t_E was shorter, while t_{PTEF} was similar while awake compared to the sleeping state (**Table 3, Figure 11**). Six (6.6%) of the 91 infants had a $t_{PTEF}/t_E < 0.25$ in the awake state, compared to 33 (36.3%) in the sleeping state (*P*-value < .001). The OR of having a ratio of $t_{PTEF}/t_E < 0.25$ was higher in the sleeping compared to the awake state; crude OR (95% CI) 10.0 (3.05, 32.8).

Table 3.	Tidal-flow volume lo	pop parameters	in the awake	and sleeping s	state at three	months
(n=91).						

	Awake state	Sleeping state	
Characteristics (n=91)	Mean (95% CI)	Mean (95% CI)	P-value*
Lung function parameters			
Number of loops included	21 (18, 24)	33 (30, 37)	<.001
Respiratory rate (RR) per minute	53 (51, 56)	38 (37, 40)	<.001
Peak tidal expiratory flow (PTEF) in ml	110 (105, 116)	84.1 (80.0, 88.2)	<.001
Time parameters			
Ratio $t_{\text{PTEF}}/t_{\text{E}}$	0.39 (0.37, 0.41)	0.28 (0.27, 0.29)	<.001
Time to PTEF (t_{PTEF}) in s	0.24 (0.23, 0.26)	0.25 (0.24, 0.26)	0.402
Expiratory time (t_E) in s	0.65 (0.61, 0.68)	0.93 (0.88, 0.98)	<.001
Volume parameters			
Volume at PTEF (V_{PTEF}) in ml	18.4 (17.2, 19.6)	14.2 (13.5, 15.0)	<.001
Expiratory volume (V_E) in ml	49.1 (46.4, 51.7)	49.7 (47.4, 51.9)	0.834
Tidal volume $(V_{\rm T})$ in ml	49.4 (46.7, 52.1)	50.5 (48.2, 52.8)	0.598

* Wilcoxon signed-rank test of paired samples

Ratio tPTEF/tE: ratio of time to peak tidal expiratory flow to expiratory time.



Figure 11. Distributions of the $t_{PTEF/tE}$ ratio in awake and sleeping tidal-flow volume loop measurements at three months (n=91).¹³⁶ The bottom and top of the boxes represent the second quartile and third quartile, and the circles are the means. The vertical lines below and above the boxes are the first and fourth quartiles.

t_{PTEF}/t_E: ratio of time to peak tidal expiratory flow to expiratory time

4.1.2 Infant lung function and preschool asthma

In 1337 children with information related to the presence or absence of preschool asthma at three years as well as either TFV loops in the awake state, TEWL, eczema at three months and/or *FLG* mutations, the association between infant lung function and preschool asthma was explored (**sub study IV**). Information related to the presence or absence of preschool asthma and infant lung function was available in 563 children, of whom 74 (13.1%) had preschool asthma. The mean \pm SD t_{PTEF}/t_E and t_{PTEF} in infancy were 0.39 \pm 0.08 and 0.20 \pm 0.05 s, respectively.

Lower lung function by lower $t_{\text{PTEF}}/t_{\text{E}}$ and shorter t_{PTEF} were associated with preschool asthma; adjusted OR (95% CI) 5.44 (2.15, 13.8) and 2.71 (1.48, 4.94), respectively (**Figure 12**).



Figure 12. Adjusted OR (95% CI) for preschool asthma at three years, according to the presence of lower t_{PTEF}/t_E , and shorter t_{PTEF} at three months (n=563). Logistic regression models adjusted for sex, GA at birth, tobacco smoke exposure in pregnancy, urban living environment in pregnancy, breastfeeding, parental allergic disease (any), parental education level and the PreventADALL interventions.

(a) Lower $t_{\text{PTEF}}/t_{\text{E}}$ equal to a $t_{\text{PTEF}}/t_{\text{E}} \le 0.25$

(b) Shorter t_{PTEF} equal to a $t_{\text{PTEF}} < 0.17$ s (<25th percentile)

*t*_{PTEF}/*t*_E: time to peak tidal expiratory flow to total expiratory time; *t*_{PTEF}: time to peak tidal expiratory flow, OR: odds ratio, GA: gestational age, PreventADALL: Preventing Atopic Dermatitis and ALLergies in children

4.2 SKIN BARRIER FUNCTION

4.2.1 Skin barrier function and infant lung function

In a study population of 899 infants, we explored whether reduced skin barrier function by high TEWL, manifestations of eczema at three months or *FLG* mutations, were associated with lower lung function at three months (**sub study II**). At three months, the mean \pm SD TEWL was 7.94 \pm 5.66 g/m²/h (n=827), while eczema was present in 135/898 (15.0%) and *FLG* mutations identified in 73/730 (10.0%) infants. Lung function by TFV loops in the awake state was available in 899 infants, with a mean t_{PTEF}/t_E and t_{PTEF} of 0.39 \pm 0.08 and 0.21 \pm 0.05 s, respectively.

Overall, no associations between high TEWL, eczema or *FLG* mutations and lower lung function ($t_{PTEF}/t_E < 0.25$) were found. However, an inverse association between high TEWL and t_{PTEF} was observed, adjusted OR (95% CI) 1.61 (1.08, 2.42) (**Figure 13**). Similar results were seen using the continuous t_{PTEF} , with a high TEWL negatively associated with t_{PTEF} ; β coefficient (95% CI) -0.01 (-0.20, -0.00), while the presence of eczema or *FLG* mutations were not.



Figure 13. Adjusted OR (95% CI) for lower $t_{PTEF}^{b)}$ at three months, according to the presence of high TEWL, eczema or FLG mutations at three months (n=899).¹³⁷ Logistic regression models adjusted for sex, GA at birth, nicotine exposure in pregnancy, parental asthma, parental education level, weight at three months and the PreventADALL skin intervention.

^(a) High TEWL equal to mean TEWL >8.83 g/m²/h (>75th percentile)^(b) Lower t_{PTEF} equal to a t_{PTEF} <0.17 s (<25th percentile) TEWL: transepidermal water loss, FLG: filaggrin, OR: odds ratio, GA: gestational age, PreventADALL: Preventing Atopic Dermatitis and ALLergies in children

4.2.2 Infant skin barrier function and preschool asthma

To assess the associations between infant skin barrier function and preschool asthma, 1337 children with information related to the presence or absence of preschool asthma at three years as well as either TEWL, eczema at three months and/or *FLG* mutations were studied (**sub study IV**). Preschool asthma was present in 180/1337 (13.5%) children. The mean \pm SD TEWL was 8.41 \pm 6.12 g/m²/h (n=1173), while 168/1318 (12.7%) had eczema at three months and 96/1045 (9.2%) carried a *FLG* mutation. In **Figure 14** and **Table 4**, the distribution of high TEWL, eczema and *FLG* mutations among the children with (n=180) and without (n=1157) preschool asthma is shown, respectively.



Figure 14. Distribution of high TEWL, eczema at three months and FLG mutations among the children with preschool asthma at three years (n=180).¹³⁸

^(a) High TEWL equals to a mean TEWL >9.50 g/m²/h (>75th percentile) TEWL: transepidermal water loss, FLG: filaggrin

Table 4. Distribution of high TEWL, eczema at three months and FLG mutations among the children without preschool asthma at three years (n=1157).

	Children <i>without</i> preschool asthma		
Characteristics	n (%)		
High TEWL (a)	244/1014 (24.1)*		
Eczema	139/1141 (12.2)		
FLG mutations	78/910 (8.57)		

* Significant difference in distribution compared to children *with* preschool asthma at 3 years (*P*-value < .05) ^(a) High TEWL equals to a mean TEWL >9.50 g/m²/h (>75th percentile)

TEWL: transepidermal water loss, FLG: filaggrin

While a high TEWL was associated with preschool asthma; adjusted OR (95% CI) 1.61 (1.06, 2.44), eczema or *FLG* mutations were not (**Figure 15**).



Figure 15. Adjusted OR (95% CI) for preschool asthma at three years, according to the presence of high TEWL, eczema at three months and FLG mutations. Logistic regression models adjusted for sex, GA at birth, tobacco smoke exposure in pregnancy, urban living environment in pregnancy, breastfeeding, parental allergic disease (any), parental education level and the PreventADALL interventions. Models including FLG mutations were only adjusted for sex.

(a) High TEWL equals to a mean TEWL >9.50 g/m²/h (>75th percentile)

TEWL: transepidermal water loss, FLG: filaggrin, OR: odds ratio, GA: gestational age, PreventADALL: Preventing Atopic Dermatitis and ALLergies in children

4.3 EARLY-LIFE EOSINOPHIL INFLAMMATION

4.3.1 Early-life eosinophil inflammation and preschool asthma

Based on 1233 children with information on EDN levels at one and/or three years (**sub study III**), the associations between EDN levels at one and three years and preschool asthma at three years were investigated. Among the children with available EDN levels at one year and three years, 68/787 (6.8%) and 76/857 (8.9%) had preschool asthma, respectively. The median EDN levels were higher at one year compared to at three years; 27.8 μ g/L versus 19.9 μ g/L (*P*-value < .001) (**Figure 16**).



Figure 16. EDN levels at one (n=787) and three (n=857) years (16.1), and in children with information on EDN at both ages (n=411) (16.2). From the bottom of the boxes, horizontal lines represent the second quartile, median and third quartile. Vertical lines below and above are the first and fourth quartile, respectively. The narrow tips represent EDN levels above the fourth quartile.

^(a) EDN levels at 1 year among all children with information on EDN at 1 year ^(b) EDN levels at 3 years among all children with information on EDN at 3 years ^(c) EDN levels at 1 year among children with information at both 1 and 3 years ^(d) EDN levels at 3 years among children with information at both 1 and 3 years ^(d) EDN levels at 3 years among children with information at both 1 and 3 years EDN: eosinophil-derived neurotoxin, $\mu g/L$: micrograms per liter

Compared to non-atopic children (no history of asthma, wheeze, doctor-diagnosed asthma, use of and/or prescribed asthma medications, AS or AD) at three years, median EDN levels at both one and three years were significantly higher in children with preschool asthma, as well as AS, AD, and any allergic disease at three years (P-values < .05), see **Table 5**.

<i>Table 5.</i> EDN levels in children with information at one $(n=787)$ and three $(n=857)$ years, at
both ages $(n=411)$, as well as EDN levels at one and three years by atopic status at three
years.

	1 year		3 years	
	Median		Median	
Characteristics	(IQR; min-max)	n	(IQR; min-max)	n
EDN levels (µg/L) by age				
EDN at 1 year	27.4 (31.7; 5.61-243)	787		
EDN at 3 years			20.1 (20.4; 4.00-304)	857
EDN at both 1 and 3 years	27.8 (32.3; 5.61-237)	411	19.9 (20.1; 4.00-235)	411
EDN levels (μg/L) by atopic status at 3 years				
Preschool asthma (a)	32.6 (44.1; 6.14-237)	68	25.1 (20.5; 7.10-304)	76
Allergic sensitization (b)	42.0 (46.9; 9.10-205)	95	28.3 (30.5; 7.15-304)	176
Atopic dermatitis (c)	35.5 (43.2; 5.85-242)	154	22.5 (21.9; 5.72-254)	201
Any allergic disease (d)	33.5 (38.3; 5.85-242)	311	22.6 (23.0; 4.00-304)	404
Non-atopic children (e)	24.0 (31.1; 5.86-216)	147	17.3 (17.0; 5.00-245)	173

^(a) Preschool asthma at 3 years was defined as having \geq 3 episodes of recurrent bronchial obstruction between 2 and 3 years, and fulfilling at least one of the following criteria; 1) doctor-diagnosed asthma between 0 and 3 years, and 2) any use of and/or prescribed asthma medications (bronchodilators, inhaled corticosteroids or leukotriene-antagonist) between 9 months and 3 years ^(b) Allergic sensitization at 3 years was defined as having positive SPT (\geq 3 mm) and/or specific IgE level (>0.35 kU_A/L) towards any allergen (egg, cow's milk, peanut, wheat, soy, cod, birch, grass, mugwort, dog, cat, horse, house dust mite and/or mold) at 3 years ^(c) Atopic dermatitis at 3 years was defined as fulfilling the United Kingdom Working Party and/or Hanifin and Rajka criteria at 3 years ^(d) Any allergic disease at 3 years was defined as children with either any wheeze, preschool asthma, allergic sensitization and/or atopic dermatitis at 3 years ^(e) Non-atopic children were defined as children with no history of any wheeze, doctor-diagnosed asthma, use of and/or prescribed asthma medications, allergic sensitization or atopic dermatitis between birth and 3 years

IQR: interquartile range, EDN: eosinophil-derived neurotoxin, $\mu g/L$: micrograms per liter, SPT: skin prick test, kU_A/L : kilounits of allergen specific IgE per liter

Compared to non-atopic children by three years, higher EDN levels at one and three years increased the OR of preschool asthma; adjusted OR (95% CI) 2.20 (1.09, 4.41) and 4.68 (2.29, 9.55), respectively (**Figure 17**).



Figure 17. Adjusted OR (95% CI) for preschool asthma at 3 years according to higher EDN levels at one and three years. Logistic regression models adjusted for sex, GA at birth, tobacco smoke in pregnancy, parental allergic disease (any), cesarean section and the PreventADALL interventions.

^(a) Higher EDN levels at 1 year by Youden's Index for preschool asthma at 3 years equal to EDN levels \geq 20.7 µg/L at 1 year ^(b) Higher EDN levels at 3 years by Youden's Index for preschool asthma at 3 years equal to EDN levels \geq 20.5 µg/L at 3 years

EDN: eosinophil-derived neurotoxin, OR: odds ratio, GA: gestational age, PreventADALL: Preventing Atopic Dermatitis and ALLergies in children

4.4 SEX DIFFERENCES

4.4.1 Infant lung- and skin barrier function, early-life eosinophil inflammation and preschool asthma by sex

Of the 1337 children included in **sub study IV**, boys had higher rates of preschool asthma at three years (n=180) compared to girls (*P*-value < .001); 122/702 (17.4%) versus 58/635 (9.1%). Similarly, in **sub study III** (n=1233), rates of preschool asthma at three years (n=109) were higher in boys compared to girls (*P*-value =0.013); 71/482 (14.7%) versus 38/410 (9.3%). Compared to girls, the t_{PTEF}/t_E and t_{PTEF} in infancy were lower in boys (**sub study IV**), while early-life EDN levels were higher (**sub study III**). The mean TEWL and rates of eczema in infancy and *FLG* mutations (**sub study IV**) were similar across the sexes (**Table 6**).

Table 6. Tidal-flow volume loop parameters (n=563), TEWL (n=1173), eczema (n=1318) at three months, FLG mutations (n=1045), as well as EDN levels at one (n=787) and three (n=857) years by sex.

	Boys	Girls	
Characteristics	Descriptive statistics	Descriptive statistics	P-value*
TFV loop parameters at 3 months			
Ratio $t_{\text{PTEF}}/t_{\text{E}}$ (n=563), mean (SD)	0.38 (0.08)	0.40 (0.08)	0.019
t_{PTEF} in s (n=563), mean (SD)	0.20 (0.05)	0.21 (0.05)	<.001
Skin conditions at 3 months			
TEWL (n=1173), mean (SD)	8.68 (5.93)	8.09 (6.32)	0.071
Eczema (n=1318), n (%)	95 (13.7)	73 (11.6)	0.244
FLG mutations (n=1045), n (%)	57 (10.2)	39 (8.0)	0.211
EDN levels (µg/L) at 1 and 3 years			
1 year (n=787), median (IQR; min-max)	32.0 (34.9; 5.95-243)	24.5 (28.1; 5.61-242)	<.001
3 years (n=857), median (IQR; min-max)	20.9 (21.9; 5.05-304)	19.0 (18.4; 4.00-235)	0.018

* Independent t-test, Chi2 test or Mann-Whitney U-test

IQR: interquartile range, TFV: tidal flow-volume, t_{PTEF}/t_E : time to peak tidal expiratory flow to total expiratory time; t_{PTEF} : time to peak tidal expiratory flow, TEWL: transepidermal water loss, *FLG*: filaggrin, EDN: eosinophil-derived neurotoxin, $\mu g/L$: micrograms per liter

While lower t_{PTEF/t_E} and shorter t_{PTEF} were associated with preschool asthma in boys; adjusted OR (95% CI) 7.76 (2.53, 23.8) and 3.21 (1.54, 6.69), a high TEWL, eczema or *FLG* mutations were not. Carrying a *FLG* mutation was associated with the development of preschool asthma in girls; adjusted OR (95% CI) 3.48 (1.48, 8.21), while lower t_{PTEF/t_E} and t_{PTEF} , high TEWL and eczema were not (**sub study IV**), see **Figure 18**.

Higher EDN levels at one year were associated with the development of preschool asthma in boys, adjusted OR (95% CI) 4.48 (1.52, 13.2), but not in girls. At three years, higher EDN levels were associated with preschool asthma in both boys and girls; adjusted OR (95% CI) 4.22 (1.87, 9.54) and 7.58 (1.82, 31.6), respectively (**sub study III**), see **Figure 18**.



Figure 18. Adjusted OR (95% CI) for preschool asthma at three years by sex, according to the presence of lower t_{PTEF}/t_E and shorter t_{PTEF} (n=563), high TEWL (n=1173), eczema (n=1318) at three months, FLG mutations (n=1045) as well as higher EDN levels at one (n=787) and three (n=857) years. Logistic regression models were adjusted for GA at birth, tobacco smoke exposure in pregnancy, urban living environment in pregnancy, breastfeeding, parental allergic disease (any), parental education level, and the PreventADALL interventions. Models including FLG mutations remained unadjusted, while models including EDN were adjusted for GA at birth, tobacco smoke exposure in pregnancy, parental allergic disease (any), cesarean section and the PreventADALL interventions.

^(a) Lower t_{PTEF}/t_E equal to a $t_{PTEF}/t_E < 0.25$ ^(b) Shorter t_{PTEF} equal to a $t_{PTEF} < 0.17$ s ($<25^{th}$ percentile)^(c) High TEWL equals to a mean TEWL >9.50 g/m²/h ($>75^{th}$ percentile)^(d) Higher EDN levels at 1 year by Youden's Index for asthma at 3 years equal to EDN levels $\geq 26.7 \ \mu g/L$ at 1 year ^(e) Higher EDN levels at 3 years by Youden's Index for asthma at 3 years equal to EDN levels $\geq 20.5 \ \mu g/L$ at 3 years

*t*_{PTEF}/*t*_E: time to peak tidal expiratory flow to total expiratory time; *t*_{PTEF}: time to peak tidal expiratory flow, TEWL: transepidermal water loss, *FLG*: filaggrin, EDN: eosinophil-derived neurotoxin, OR: odds ratio, GA: gestational age, PreventADALL: Preventing Atopic Dermatitis and ALLergies in children

5 DISCUSSION

5.1 MAIN FINDINGS

The overall aim of this doctoral thesis was to explore early risk factors for the development of preschool asthma, related to infant lung function, inflammation and skin barrier function in early life. The main findings of sub studies I-IV are illustrated in **Figure 19**.

First, infant lung function differed by arousal state, with overall higher TFV loop values while awake compared to the sleeping state. Both lower t_{PTEF}/t_E and shorter t_{PTEF} while awake in early infancy were positively associated with preschool asthma. Second, overall, neither eczema nor *FLG* mutations were associated with infant lung function or preschool asthma. However, positive associations between high TEWL and shorter t_{PTEF} as well as preschool asthma were seen. Third, overall, higher EDN levels in infancy and at preschool age were associated with preschool asthma.

Last, infant lung function differed by sex, with lower t_{PTEF}/t_E and t_{PTEF} observed in boys compared to girls, whereas in relation to skin barrier function, similar rates of TEWL, eczema and *FLG* mutations were seen. Regarding early-life inflammation, EDN levels were higher in boys compared to girls in both infancy and at preschool age. Preschool asthma was more frequent in boys. The associations to asthma differed according to sex, with both lower t_{PTEF}/t_E and shorter t_{PTEF} in infancy observed in boys and *FLG* mutations seen in girls only. Neither high TEWL nor eczema in early life was associated with preschool asthma when stratifying for sex. Higher EDN levels in infancy were only associated with preschool asthma in boys, whereas at preschool age, positive associations were seen among both sexes. Interestingly, at preschool age, the highest OR of preschool asthma were observed in in girls.



Figure 19. Figure of the main findings of the four sub studies (I-IV) included in this thesis, with 19.1 illustrating that infant lung function at three months differed between awake and sleeping arousal state, as well as infant lung function in the awake state at three months being associated with preschool asthma at three years. A high TEWL at three months was associated with lower infant lung function at three months and preschool asthma at three years (19.2), while higher EDN levels at one and three years overall were associated with future as well as concurrent preschool asthma at three years (19.3). The associations to preschool asthma at three years differed by sex, with infant lung function at three months being associated with preschool asthma in boys, while FLG mutations were linked to preschool asthma in girls only (19.4).

TEWL: transepidermal water loss, EDN: eosinophil-derived neurotoxin, FLG: filaggrin

* The illustration was created with Biorender.com

5.2 MAIN FINDINGS IN RELATION TO PREVIOUS RESEARCH

5.2.1 Infant lung function

5.2.1.1 Infant lung function and arousal state

Our findings of higher mean t_{PTEF}/t_E while awake compared to the sleeping state is in line with a study from the 1990's including 19 healthy Norwegian newborns.¹³⁹ Apart from the latter study, we are unaware of other studies assessing whether TFV loops differ according to arousal state in infants, older children and adults. Comprising the t_{PTEF}/t_E ratio, t_E was longer in the sleeping state, while t_{PTEF} was similar. During sleep, skeletal muscle tone is lower and may necessitate increased diaphragmic work of breathing,¹⁴⁰ which possibly can explain the longer expiratory time required (t_E) while sleeping. In support of this, one study of 70 sedated infants and children observed t_{PTEF} , t_E and t_{PTEF}/t_E varying during sleep stages, likely linked to lower tone of respiratory muscles.¹⁴¹ In addition, breathing during sleep is disposed to airway obstruction,¹⁴² with asthma symptoms commonly deteriorating,¹⁴³ possibly reflected in the lower t_{PTEF}/t_E observed while sleeping. Though all TFV loops were recorded with infants in a supine position with tidal volumes (V_T) remaining similar across the arousal states, the generally lower respiratory rates (*RR*) may have contributed to the lower t_{PTEF}/t_E while sleeping, supported by a previous paper of 33 sedated infants emphasizing the positive correlation between *RR* and t_{PTEF}/t_E .¹⁴⁴

5.2.1.2 Infant lung function and preschool asthma

Lower t_{PTEF}/t_E in infancy was associated with the development of preschool asthma, in line with similar studies.^{59,145} However, Guerra *et al.* used a different technique to measure lung function (forced expiratory flow-volume maneuver) among more than 100 sleeping and/or sedated American infants, and asthma was defined by doctors diagnose received at any time point between six and 36 years, reported in questionnaires.¹⁴⁵ In relation to asthma-like symptoms, a previous Norwegian study found an association between lower t_{PTEF}/t_E and wheezing at preschool age,¹²⁸ supporting our findings. The shorter t_{PTEF} in infancy positively associated with preschool asthma is novel. Previous studies have mainly found lower t_{PTEF} values in infants and children with airway obstruction, but support our observations.^{66-68,146}

5.2.2 Infant skin barrier function

5.2.2.1 Skin barrier function and infant lung function

From a large population-based pediatric population, we are first to report an observation between high TEWL and infant lung function, indicating a potential link between the two in early life. However, as the associations to shorter and continuous t_{PTEF} only were significant in adjusted models, our observations may possibly be chance findings. In respect to asthma, a German study of 95 adolescents and adults found no difference in TEWL values between individuals with allergic asthma and controls.¹⁴⁷ Though no association between TEWL and $t_{\text{PTEF}/\text{TE}}$ was observed, a high TEWL was associated with a shorter time before reaching peak

expiratory flow (t_{PTEF}), a characteristic frequently observed in infants and children with airway obstruction.⁶⁶⁻⁶⁸

Though both lower lung function and AD in early life are linked to the development of preschool asthma,^{36,59} we found no association between eczema and $t_{\text{PTEF}}/t_{\text{E}}$ or t_{PTEF} , corresponding with earlier findings.^{148,149} Hu *et al.* reported no difference in lung function by spirometry across eczema phenotypes in 4227 school-aged Dutch children.¹⁴⁸ Similarly, in a study of 135 Norwegian children, Håland *et al.* concluded that lung function by TFV loops appeared independent of AD.¹⁴⁹ Yet, evidence suggests the onset and severity of AD may be connected to asthma development,¹⁵⁰ factors beyond the scope of my research. In the Norwegian paper by Lødrup Carlsen *et al.*, lung function from birth (TFV loops) to late adolescence (spirometry) was tracked in 329 children who since infancy had developed allergic asthma and AD.⁶²

Though no associations between *FLG* mutations and t_{PTEF}/t_E or t_{PTEF} in early infancy were seen, our findings are an important contribution to the research field. In regards to airway obstruction, a large British study including 2312 infants observed an association between *FLG* mutations and wheeze in early infancy.¹⁵¹ Though *FLG* mutations are hypothesized to contribute to asthma,³⁶ the risk is likely highest in children with a previous AD diagnosis.^{71,77}

5.2.2.2 Infant skin barrier function and preschool asthma

The significant association between a high TEWL in infancy and preschool asthma, hitherto not reported, provide support to the hypothesized role of the skin barrier in asthma development.⁷⁴ To our knowledge, the only paper with a similar objective was performed by Löffler *et al.* (n=95), reporting no differences in TEWL values using a patch test across 25 subjects with allergic asthma and 26 healthy controls aged 14 and older, recruited from outand in-patient clinics in Germany.¹⁴⁷ However, with several differences in methodology and study population, the results reported in the latter study¹⁴⁷ may not be comparable to our findings.

Overall, no associations between eczema in infancy or *FLG* mutations and preschool asthma were seen in the absence of AD or AS, in line with several previous studies.¹⁵²⁻¹⁵⁴ Thus, our results suggest asthma with allergic comorbidities is more likely to represent an asthma phenotype persisting beyond preschool age.¹⁵⁵ With an itchy rash present at enrollment, a large Swedish report including 3124 children observed a two-fold risk of developing asthma,¹⁵⁶ contradictory to our observations. However, children were older (1-2 years) at baseline and follow-up was performed later in childhood (5 years after recruitment). In addition, a different definition of eczema was used and was reported by parents.¹⁵⁶ To the best of our knowledge, evidence of a direct association between *FLG* mutations and asthma independent of allergic comorbidities is lacking, as confirmed by our study.

5.2.3 Early-life eosinophil inflammation

5.2.3.1 Early-life eosinophil inflammation and preschool asthma

To our knowledge, we are first to investigate the relationship between EDN levels at one year and future preschool asthma, as well as EDN levels at three years and concurrent preschool asthma, demonstrating a positive correlation between higher EDN levels at both ages and preschool asthma. In line with our findings, Remes *et al.* reported significant associations between EDN as well as ECP levels and concurrent asthma in school-aged Finnish children, compared to healthy controls.¹⁰¹ In contrast to our study, the latter study population was smaller (n=235), heterogeneous in terms of age (7-12 years) and asthma was parental-reported,¹⁰¹ and may thereby not be comparable to our population. Nonetheless, several studies have reported higher EDN levels in children with concurrent asthma,^{101,105-111} associated with bronchial hyper-responsiveness,^{107,108} higher respiratory symptom burden^{106,107,112} and disuse of anti-inflammatory medications,¹⁰¹ supporting our findings. In relation to wheeze among 200 South Korean infants and toddlers recovering from bronchiolitis, Kim CK *et al.* reported a positive association between EDN levels and future wheezing episodes, further supporting our findings.¹¹⁵

5.2.4 Sex differences

5.2.4.1 Infant lung- and skin barrier function, early-life eosinophil inflammation and preschool asthma by sex

The associations between TFV loops in healthy, awake infants and preschool asthma differed by sex, with lower t_{PTEF}/t_E and shorter t_{PTEF} in infancy observed in boys compared to girls, are novel findings. However, our results are in line with the tendency of lower lung function in young boys observed in previous cohort studies.¹⁵⁷ Further supporting our findings, in 56 British infants born preterm, Stocks *et al.* reported lower t_{PTEF}/t_E and t_{PTEF} in boys compared to girls in the natural, unsedated sleeping state using the multiple occlusion technique.¹⁵⁸ In contrast, a larger Norwegian study (n=803) reported higher t_{PTEF}/t_E in the awake state in newborn boys compared to girls.⁵⁷ Compared to girls, the narrower airways relative to lung size in early life in boys¹¹⁷ may contribute to the lower TFV loop parameters observed in boys. In addition, both male sex and lower infant lung function are known risk factors for asthma.^{59,65} In line with the global trend in young children,⁴ boys had higher rates of preschool asthma compared to girls (17.4% versus 9.1%) in our study. All things considered, both differences in lung structure and asthma prevalence at preschool age between the sexes may help explain our findings.

In regard to early skin barrier function, FLG mutations were associated with the development of preschool asthma in girls only, whereas neither high TEWL nor eczema remained significant risk factors when stratifying for sex. To our knowledge, we are first to report a sex-dependent association between FLG mutations and preschool asthma. In relation to AD, a previous Russian study (n=1312) found a positive link between single nucleotide polymorphisms in the FLG gene in adult women, but not men.¹⁵⁹ We are unaware of previous publications assessing the association between TEWL and preschool asthma by sex, wherefore we provide new knowledge. The non-significant association between eczema and preschool asthma observed in both girls and boys differs from the findings in a previous Australian publication investigating 620 infants with a family history of allergic diseases, in which eczema during the first two years of life increased the risk of preschool asthma in boys.¹⁶⁰ Apart from that the latter study included infants at higher risk of developing allergic diseases,¹⁶⁰ one possible explanation is that we only focused on eczema at three months, a single time point in early infancy where an association might be harder to detect. Further, of the children identified with eczema in early life, many may not progress to develop AD in later infancy.

Concerning eosinophil-inflammation in early life, in both infancy and at preschool age, serum EDN levels were higher in boys compared to girls and is in line with a previous study of adults.¹⁶¹ In the Finnish study (n=235) by Remes et al., no significant differences in EDN or ECP levels were observed between the sexes, though school-aged boys tended to have higher serum EDN levels.¹⁰¹ Similarly, in two population-based studies of 216 and 968 school-aged children and adolescents in Norway and Australia, respectively, no differences in serum ECP levels were found.96,97 Helping to understand the observed sex differences in our study, EDN correlates to eosinophil count,^{109,161-163} and male sex is linked to higher eosinophil counts.¹⁶⁴ Further, we are first to report that higher EDN levels in infancy were positively associated with preschool asthma in boys but not girls, whereas higher EDN levels at preschool age were linked to concurrent preschool asthma among both sexes, with the highest OR observed in girls although rates of preschool asthma were higher in boys (14.7%) compared to girls (9.3%). The latter finding differs to the Swedish report of Mogensen *et al.* (n=403) with similar objective, demonstrating the highest risk of having fixed airflow obstruction by spirometry in males with elevated serum ECP levels, compared to girls.¹⁶⁵ However, with only adult participants (17-75 years), all with self-reported doctor-diagnosed asthma and/or chronic rhinosinusitis,¹⁶⁵ their results may not be comparable to our findings.

5.3 STRENGTHS AND LIMITATIONS

The four sub studies included in this thesis have several mutual strengths. First, all sub studies are based on information prospectively gathered from children and mothers participating in the PreventADALL birth cohort, antenatally enrolled from the general population. Further are the children clinically well-characterized, with detailed information on birth and background data as well as exposures and outcomes, increasing the generalizability of our findings. However, with slightly higher education level among the parents in the cohort compared to the general populations. To increase the inter-observer reliability, health care professionals were trained together, and the study centers in Norway and Sweden used the same standard operating procedures to assess lung function, TEWL and eczema.

Based on a power analysis performed prior to data collection, the population size was sufficient to compare TFV loop parameters between paired awake and sleeping lung function

measurements in 91 infants the different arousal states. Exploring the association between skin barrier function and infant lung function (n=899), the high numbers of infants with information on TFV loops (100%) as well as TEWL (91.9%), eczema (99.9%) and/or *FLG* mutations (81.2%) are a strength, as a larger study size may reduce the risk of type II errors; when researchers fail to reject the null hypothesis when it is actually false.¹⁶⁶ Similarly, when assessing the correlations between infant lung function, skin barrier function and preschool asthma (n=1337), the high number of children with information on preschool asthma (100%) as well as TEWL (87.7%), eczema (98.6%) and/or *FLG* mutations (78.2%) are a strength, whilst fewer children had information on TFV loops (42.1%).

Although TFV loops, as a non-invasive technique, is an asset in epidemiological research, questions regarding the possible influence of conditions such as arousal state, positioning, respiratory rate and age have earlier been raised.¹⁶⁷ The process of recording and analyzing has also been discussed, in which the difficulties of obtaining reproducible TFV loops has been a concern.⁶⁷ Aiming to control for some of these issues, standard operating procedures to measure as well as manually assess the quality of infant lung function measurements were used,¹²³ attaining a large material on TFV loops from healthy infants in the awake (n=899) and sleeping (n=375) state, collected at a specific time point under similar conditions by trained investigators. Further, TFV loops were obtained in early infancy, limiting the risk of postnatal adverse events influencing lung function development. While several studies have demonstrated that a $t_{\text{PTEF}}/t_{\text{E}} < 0.25$ in early life is associated with the development of asthma, 59,128-130 no clear cut-off for shorter t_{PTEF} has previously been determined and is a limitation. However, as lower values have been reported in infants and children with airway obstruction, $^{66-68}$ a t_{PTEF} in the lower range (<25th percentile) was chosen. Lung function was not measured in the less densely populated county of Østfold (Norway), where socioeconomical status in general was lower compared to the capital cities Oslo (Norway) and Stockholm (Sweden),¹⁶⁸ and limits the generalizability of our observations to similar populations.

By including TEWL measurements to objectively assess skin barrier function and using established cut-offs (>75th percentile) we strengthen our findings.^{126,131} Choosing physicianconfirmed eczema instead of AD fulfilling the UKWP criteria in early infancy might be considered as a limitation. However, the major UKWP AD criteria of an itchy rash has been questioned in a previous PreventADALL study, showing less than one fifth of the children with eczema at three months later diagnosed with AD by one year fulfilled this requirement in early infancy.¹³² In support, advanced motor skills are needed to locate an itchy target on the body, an ability which may not yet fully developed in early infancy.¹⁶⁹ The rate of *FLG* mutation carriers in the sub studies (9.0-10.0%) corresponds with the prevalence in the PreventADALL study as well as in European populations.¹²⁷

To the best of our knowledge, we are first to provide population-based serum EDN levels retrieved from a large pediatric population at two specific time points in childhood. Though previous studies foremost analyzed serum EDN with manual enzyme-linked immunosorbent assays,¹⁷⁰ the automated ImmunoCAP EDN research assay used appears to be a reliable complement with high precision.¹⁰⁹ As infections in addition to allergens can trigger the secretion of EDN from eosinophils,⁹⁰ the exclusion of data on infections is a limitation, but was beyond the scope of my research. With eosinophil counts not available, we are also unable to elaborate on previous reports of correlations between eosinophil counts and EDN^{109,161-163} or sex,¹⁶⁴ and is a limitation to our findings.

While wheeze is the most commonly occurring symptom of preschool asthma, all wheezing is not related to asthma⁵ wherefore incidence, severity and symptom patterns must therefore be taken into consideration in diagnosis at this age.¹⁰ To capture the incidence and symptom pattern, rBO (\geq 3 episodes of wheeze) in the recent year was a mandatory criterion, together with a history of doctor-diagnosed asthma and/or asthma medication use,¹³³ attempting to create a strict definition for preschool asthma. However, in many children preschool asthma-like symptoms resolve with age,¹⁷¹ while other children experience the first manifestations of chronic asthma at preschool age, pointing to the uncertainty of diagnosing asthma at this age. In addition to respiratory symptoms such as rBO, most guidelines therefore advise basing possible asthma diagnosis on response to asthma medications and the exclusion of differential diagnoses,¹¹ information which was not available to evaluate and is a limitation. Nevertheless, a recent birth cohort study reported a 20-150-fold higher risk of asthma at five years among children experiencing wheeze reported between three months and five years.¹² It would however be interesting to assess the associations between infant lung function, skin barrier function, EDN and asthma in children with persisting symptoms at an older age.

In none of the included sub studies were significant interaction effects between the PreventADALL skin or food intervention groups and the exposures on the outcomes observed, wherefore all children were included in our analyses. To reduce the possibility that effect-modifying factors, such as the PreventADALL interventions, heredity for allergic diseases or sex, had influence on our findings, based on the literature we adjusted for possible confounders. The majority of the potential risk factors used to adjust multivariate regression analyses were recorded prior to the clinical investigation from which all results are presented, and is a strength. Further, when applicable, continuous variables were used in addition to dichotomized variables, such as TFV loop parameters and TEWL.

6 CONCLUSIONS

Based on the results from the four sub studies (I-IV) included in this thesis, we conclude that both infant lung- and skin barrier function as well as early-life eosinophil inflammation may be linked to the development of preschool asthma, in which sex appears to influence (**Figure 20**).

Infant lung function

- The *t*_{PTEF}/*t*_E differed between the arousal states, with overall higher TFV loop values while awake compared to sleeping state. TFV loops should be analyzed and interpreted according to arousal state, using separate reference values.
- Lower *t*_{PTEF}/*t*_E and shorter *t*_{PTEF} in the awake state in early infancy were associated with prechool asthma.

Infant skin barrier function

• High TEWL was inversely associated with *t*_{PTEF} in early infancy as well as the development of preschool asthma, indicating a possible link between the skin barrierand lung function in early infancy as well as later preschool asthma.

Early-life eosinophil inflammation

• We propose population-based serum EDN reference values, displaying higher EDN levels in infancy compared to at preschool age. Higher EDN levels at both ages were associated with preschool asthma, indicating early-life eosinophil inflammation may play a role in preschool asthma development.

Sex differences

- Associations to preschool asthma differed by sex, with lower *t*_{PTEF}/*t*_E and shorter *t*_{PTEF} observed in boys only.
- *FLG* mutations were associated with preschool asthma in girls only, while neither eczema or high TEWL remained significant risk factors in either sex.
- Higher EDN levels in infancy were associated with preschool asthma in boys only. At preschool age, higher EDN leveles were associated with concurrent preschool asthma among both sexes, with the highest OR observed in girls.



Figure 20. In summary, lower infant lung function, skin barrier impairment as well as eosinophil inflammation in early life may be linked to the development of preschool asthma, in which sex appears to influence.

* The illustration was created with Biorender.com

7 POINTS OF PERSPECTIVE

7.1 CLINICAL IMPLICATIONS

While it is possible to obtain objective lung function tests from awake children actively participating, it has historically been challenging assessing lung function in non-sedated infants.⁵⁰ To our knowledge, we provide the largest material on TFV loops in the awake and/or natural sleeping state (n=1183), demonstrating that lung function measurements in both arousal states in early infancy are feasible, but ought to be assessed individually using separate reference values. However, lung function techniques allowing for repeated measurements in awake young children are warranted,⁴⁹ favoring measuring TFV loops in the awake state. We further suggest that lower t_{PTEF}/t_E and shorter t_{PTEF} while awake may be useful in the assessment of lung function and preschool asthma. Reflected in the overall lower t_{PTEF}/t_E observed while sleeping compared to the awake state, one may speculate that different cut-off levels of t_{PTEF}/t_E according to arousal state are needed. With t_{PTEF} remaining stable across awake and sleeping TFV measurements, t_{PTEF} may prove useful in future studies in which lung function with mixed arousal states are evaluated, though this is beyond the scope of the current research project.

The inverse associations between high TEWL and t_{PTEF} in infancy suggest a potential link between the skin barrier- and lung function in early life. From this significant finding, we additionally bring evidence of a possible association between high TEWL and preschool asthma. In summary, our significant findings suggest a possible link between early-life skin barrier impairment and respiratory health. In research and possibly clinical practice, TEWL may not only prove useful to identify infants at risk of AD⁷⁸ and AS,¹³¹ but possibly also preschool asthma. Overall, no associations between eczema in infancy or *FLG* mutations and infant TFV loops or preschool asthma were observed.

Retrieved from a large pediatric general population, we suggest the documented EDN levels may be considered as reference values in similar low-risk populations, showing higher levels in infancy compared to at preschool age, in line with the ECP levels reported in a study of 245 Norwegian children.¹⁷² Although higher EDN levels at both ages were observed in children with preschool asthma as well as AS, AD and any allergic disease compared to non-atopic children (n=235).¹⁰¹ Still, we observed that higher EDN levels at both ages positively correlated to preschool asthma. It is therefore more likely that EDN, in combination with clinical characteristics and medical history, may help assist clinicians to identify children at higher risk of preschool asthma,¹⁷³ rather than single-handedly be of diagnostic or prognostic value in pediatric clinical practice. In relation to existing biomarkers reflecting a Th2-driven inflammation, all face difficulties.^{174,175} Being easily-obtained in various body tissues such as in saliva, urine and blood, requiring small sample volumes and with stable concentrations after collection, EDN may have a potential in pediatric clinical practice.⁹³

In infancy, $t_{\text{PTEF}}/t_{\text{E}}$ and t_{PTEF} were lower in boys compared to girls, while TEWL, eczema and *FLG* mutations were similar between the sexes. Levels of EDN in both infancy and at preschool age were higher in boys than girls. Stratifying for sex, lower $t_{\text{PTEF}}/t_{\text{E}}$ and shorter t_{PTEF} were associated with preschool asthma in boys, while *FLG* mutations positively correlated to preschool asthma in girls, only. The several differences in rates and the sex-dependent associations observed suggest sex ought to be taken into consideration.

7.2 FUTURE RESEARCH

With some exceptions, developmental factors for asthma from early infancy to later childhood are not well understood.⁴⁴ Observing possible links between infant lung- and skin barrier function as well as early-life eosinophil inflammation and preschool asthma, in a time period rarely studied, future research in older children and other study populations is warranted to confirm or contradict our findings. Further, no attempts to investigate the possible underlying mechanisms of our observations were made, which overall require more evidence.

Infant lung function

- As possible effect-modifying factors of TFV loops may balance out with age,¹⁷⁶ such as respiratory rate,¹⁷⁷ future research is needed to determine whether TFV loops differ between the awake and sleeping arousal state in older children.
- While we reported an association between shorter t_{PTEF} in infancy and preschool asthma, a TFV loop parameter seldomly used, previous research has mainly described lower t_{PTEF} values in infants and children with airway obstruction.^{66-68,146} For this reason, future longitudinal studies are required to determine if t_{PTEF} while awake, sleeping, or possible mixed arousal states, may be relevant for the identification and prediction of infants and older children at risk of developing asthma.

Infant skin barrier function

• Though an inverse association between high TEWL and *t*_{PTEF} was observed in infancy, and high TEWL as well as shorter *t*_{PTEF} correlated to preschool asthma, additional research in older children is needed to detangle the potential relationship between the skin barrier- and lung function in the development of asthma. Similarly, the role of TEWL in asthma development also remains to be determined.

Early-life eosinophil inflammation

 With overlapping ranges in serum EDN levels in infancy and at preschool age between children with preschool asthma and non-atopic children, but with a two-fold higher OR of developing preschool asthma among children with higher levels in infancy, the role of EDN as a biomarker for asthma in pediatric clinical practice remains undetermined. To document reference values and assess associations to asthma, future prospective longitudinal birth cohort studies are warranted including samples from other body tissues, collected at various ages over time in childhood. • It would also be valuable to longitudinally evaluate the relationship between EDN levels and eosinophil counts as well as other markers of eosinophil inflammation. Further, the associations between lung function in early infancy and EDN levels in late infancy, as well as EDN and concurrent and future lung function, would be interesting to assess.

Sex differences

• As associations to preschool asthma differed by sex, with lower $t_{\text{PTEF}}/t_{\text{E}}$ and shorter t_{PTEF} as well as higher EDN levels in infancy seen in boys and *FLG* mutations in girls, only, it would be interesting to investigate if the same sex-dependent associations remained significant at an older age and in other pediatric populations retrieved from the general population.

8 ACKNOWLEDGEMENTS

First, to my main supervisor Björn Nordlund, thank you for your time, patience, commitment, and support. I am truly grateful to you for giving me this opportunity.

Second, I would like to thank all of my co-supervisors, Cilla Söderhäll, Karin C. Lødrup Carlsen and Catarina Almqvist, for interesting discussions, guidance, and support. To Cilla, thanks for following me on this long journey, listening and giving calming advice, while introducing me to the field of genetics. Karin, your expert knowledge, experience, structure and writing skills are inspiring, and have helped me to develop as a researcher. Thank you, Catarina, for always seeing to my best while looking after my own research interests.

To Gunilla Hedlin, thank you for your endless support and ongoing interest in my education and progress.

My mentor, Erebouni Arakelian, thank you for the inspiration, encouragement, advice, and kind support.

Many thanks to the participating families in the PreventADALL study, your contribution to research is invaluable!

Thanks to the amazing PreventADALL team in Sweden and Norway. To my former colleagues Karina Björstad Barhag and Ellen Tegnerud, you are both very skilled and it was a privilege to work with you. Sabina Wärnberg Gerdin, Anna Asarnoj, Sandra Götblad, Nora Nilsson, Natasha Sedergen, Hilde Aaneland and Anine Lie, all involved in the PreventADALL study, thanks for all your help and support.

To Karolinska Institutet and the department of Women's and Children's health, in particular Caroline Rådestad and Jennifer Frithiof, thanks for the administrative support. I would also like to thank the PhD friends and colleagues in SFO-V 2018, especially Siri Lilliesköld for our lunches, the coffee, and being my good friend.

My dear PhD colleagues, Karen Eline S. Bains and Hrefna K. Gudmundsdóttir, thank you for your friendship and an amazing teamwork! I cannot in words express my appreciation for your tireless effort helping me to reach this far. You're next!

Angela Hoyer, my friend, I am so glad we got to know each other. Thank you for the collaboration, the good times, as well as listening to my concerns and providing support when I needed it. I'm looking forward to your upcoming dissertation!

My friend and former PhD colleague, Caroline Aleksi-Olsson Mägi, thanks for the long talks, the support and for always rooting for me!

To my friends and PhD colleagues Sandra Tedner, Henrik Hjelmgren and several others, thanks for the beer, much needed conversations, and encouragement shared over drinks and dinners.

To everyone participating in the PreventADALL journal club, thank you for the interesting discussions.

To my co-authors, thank you for the inspiring and excellent feedback.

Thank you, Magnus P. Borres, Robert Movérare and Tina Ekenkrantz at Thermo Fisher Scientific for the fun and fruitful collaboration, your advice and support!

To my colleagues at the Department of Surgical Sciences, in particular Eva Ånnhagen and Camilla Fröjd, many thanks for the support and adapting my workload at Uppsala University according to my doctorial education.

My childhood friends, Anna Appelbrant, Annica Lenngren, Cicci Carstedt and Elin Åström, thank you for being my friends, always having my back, and for all the fun we've had together over the years!

To my friends, Anders Nolkrantz, Maggie Man, Helena Svernell, Julia Lindblom, Maria Qvarnström, Karin Rudberg, Hanna Arvidsson, Maria Knutsson, Anna Stefansdotter, Lovisa Isaksson, Ellie Nylander and Christina Lindahl, thank you for the phone calls, texts, the coffee, dinners and drinks, taking me out dancing, singing karaoke, and giving me a break when I needed it.

To Marie and Fredrik Bagge, Sofia Nilsson and Heiko Wilken at Åsunda, many thanks for the friendship as well as watering our plants and looking after our house when we were away.

To Eva-Lena Annebäck, thanks for the dog sitting, for giving our best friend a best friend, as well as for keeping us good company on our walks in the park.

To my parents, Ann and Thomas Färdig, thank you for believing in me, your never-ending support and encouragement. Without you, this would never have been possible! To my siblings, Kristina and Anders Färdig, thanks for the support and always being on my side. To the partners of my siblings, Chris Lindberg and Yuka Otsuki, and my little niece Sophia Färdig, thanks for the get-togethers, the babysitting and fun play. To my close relatives, in particular my grandparents Inga-Britt and in memorian Göran Färdig, thanks for the genuine interest and support in my education.

To my partner, Anders Sandström, thank you for always being there for me, through fail and foul, your support means the world! To his family, in particular Karin and in memorian Ulf Sandström, many thanks for your support and interest in my education. And little Ture, our stubborn Dachshund, thanks for dragging me out on walks, helping me take my mind off the work.

9 FINANCIAL SUPPORT

The doctorial education, sub studies and thesis would not have been possible without the generous support from financial contributors and scholarships.

In particular, I would like to thank SFO-V Karolinska Institutet, The Swedish Heart-Lung Foundation, Kerstin Hejdenbergs minnesfond, H.K.H. Kronprinsessan Lovisas förening för barnasjukvård, as well as Thermo Fisher Scientific for supporting PreventADALL with the EDN analyses.

The PreventADALL study has received funding from the following sources: South-Eastern Norway Regional Health Authority, The Norwegian Research Council, Oslo University Hospital, The University of Oslo, Health and Rehabilitation Norway, The Foundation for Healthcare and Allergy Research in Sweden -Vårdalstiftelsen, The Swedish Asthma- and Allergy Association's Research Foundation, The Swedish Research Council - the Initiative for Clinical Therapy Research, The Swedish Heart-Lung Foundation, SFO-V Karolinska Institutet, Østfold Hospital Trust, The European Union (MeDALL project), by unrestricted grants from the Norwegian Association of Asthma and Allergy. The Kloster foundation, Thermo Fisher Scientific, Uppsala, Sweden (through supplying allergen reagents) and Fürst Medical Laboratory, Oslo, Norway (through performing IgE analyses), Norwegian Society of Dermatology and Venerology, Arne Ingel's legat, Region Stockholm (ALF-project and individual grants), Forte, Swedish Order of Freemasons Foundation Barnhuset, The Sven Jerring Foundation, The Hesselman foundation, The Magnus Bergwall foundation, The Konsul Th C Bergh's Foundation. The Swedish Society of Medicine. The King Gustaf V 80th Birthday Foundation, KI grants, The Cancer- and Allergy Foundation, The Pediatric Research Foundation at Astrid Lindgren Children's Hospital, The Samaritan Foundation for Pediatric research, Barnestiftelsen at Oslo University Hospital, Roche, The Frithjof Nansen Institute.
10 REFERENCES

1. Carraro S, Scheltema N, Bont L, Baraldi E. Early-life origins of chronic respiratory diseases: understanding and promoting healthy ageing. *Eur Respir J*. 2014 Dec;44 (6):1682-96.

2. Kubo M. Innate and adaptive type 2 immunity in lung allergic inflammation. *Immunol Rev.* 2017 Jul;278 (1):162-172.

3. Peters RL, Koplin JJ, Gurrin LC, Dharmage SC, Wake M, Ponsonby AL, et al. The prevalence of food allergy and other allergic diseases in early childhood in a populationbased study: HealthNuts age 4-year follow-up. *J Allergy Clin Immunol*. 2017 Jul;140 (1):145-53.

4. Papi A, Brightling C, Pedersen SE, Reddel HK. Asthma. *Lancet*. 2018 Feb 24;391 (10122):783-800.

5. Patadia MO, Murrill LL, Corey J. Asthma: symptoms and presentation. *Otolaryngol Clin North Am.* 2014 Feb;47 (1):23-32.

6. Trivedi M, Denton E. Asthma in Children and Adults-What Are the Differences and What Can They Tell us About Asthma? *Front Pediatr.* 2019 7:256.

Duijts L. Fetal and infant origins of asthma. *Eur J Epidemiol.* 2012 Jan;27 (1):5-14.

8. von Kobyletzki LB, Beckman L, Smeeth L, McKee M, Quint JK, Abuabara K, et al. Association between childhood allergic diseases, educational attainment and occupational status in later life: systematic review protocol. *BMJ Open*. 2017 Oct 11;7 (10):e017245.

9. Mastrorilli C, Posa D, Cipriani F, Caffarelli C. Asthma and allergic rhinitis in childhood: what's new. *Pediatr Allergy Immunol.* 2016 Dec;27 (8):795-803.

10. Global Initiative for Asthma (GINA). Global Strategy for Asthma Management and Prevention. Accessed 9 October, 2020. https://ginasthma.org/wpcontent/uploads/2020/06/GINA-2020-report_20_06_04-1-wms.pdf

11. Moral L, Vizmanos G, Torres-Borrego J, Praena-Crespo M, Tortajada-Girbés M, Pellegrini FJ, et al. Asthma diagnosis in infants and preschool children: a systematic review of clinical guidelines. *Allergol Immunopathol (Madr)*. 2019 Mar-Apr;47 (2):107-121.

12. Dai R, Miliku K, Gaddipati S, Choi J, Ambalavanan A, Tran MM, et al. Wheeze trajectories: Determinants and outcomes in the CHILD Cohort Study. *J Allergy Clin Immunol.* 2022 Jun;149 (6):2153-2165.

13. Kaiser SV, Huynh T, Bacharier LB, Rosenthal JL, Bakel LA, Parkin PC, et al. Preventing Exacerbations in Preschoolers With Recurrent Wheeze: A Meta-analysis. *Pediatrics*. 2016 Jun;137 (6):e20154496.

14. Chung HL. Diagnosis and management of asthma in infants and preschoolers. *Clin Exp Pediatr.* 2022 Dec;65 (12):574-584.

15. Patel SJ, Teach SJ. Asthma. *Pediatr Rev.* 2019 Nov;40 (11):549-567.

16. Szefler SJ, Chipps B. Challenges in the treatment of asthma in children and adolescents. *Ann Allergy Asthma Immunol.* 2018 Apr;120 (4):382-388.

17. Poddighe D, Brambilla I, Licari A, Marseglia GL. Omalizumab in the Therapy of Pediatric Asthma. *Recent Pat Inflamm Allergy Drug Discov.* 2018 12 (2):103-109.

18. Brusselle GG, Koppelman GH. Biologic Therapies for Severe Asthma. *N Engl J Med.* 2022 Jan 13;386 (2):157-171.

19. Laubhahn K, Phelan KJ, Jackson DJ, Altman MC, Schaub B. What Have Mechanistic Studies Taught Us About Childhood Asthma? *J Allergy Clin Immunol Pract*. 2023 Mar;11 (3):684-692.

20. Just J, Bourgoin-Heck M, Amat F. Clinical phenotypes in asthma during childhood. *Clin Exp Allergy*. 2017 Jul;47 (7):848-855.

21. Ullemar V, Magnusson PK, Lundholm C, Zettergren A, Melén E, Lichtenstein P, et al. Heritability and confirmation of genetic association studies for childhood asthma in twins. *Allergy*. 2016 Feb;71 (2):230-8.

22. Bobolea I, Arismendi E, Valero A, Agustí A. Early Life Origins of Asthma: A Review of Potential Effectors. *J Investig Allergol Clin Immunol.* 2019 29 (3):168-179.

23. Svanes C, Holloway JW, Krauss-Etschmann S. Preconception origins of asthma, allergies and lung function: The influence of previous generations on the respiratory health of our children. *Journal of Internal Medicine*. 2023 293 (5):531-549.

24. Gruzieva O, Merid SK, Melen E. An update on epigenetics and childhood respiratory diseases. *Paediatr Respir Rev.* 2014 Dec;15 (4):348-54.

25. Propp P, Becker A. Prevention of asthma: where are we in the 21st century? *Expert Rev Clin Immunol.* 2013 Dec;9 (12):1267-78.

26. Kopel LS, Phipatanakul W, Gaffin JM. Social disadvantage and asthma control in children. *Paediatr Respir Rev.* 2014 Sep;15 (3):256-62.

27. Appleton AA, Holdsworth EA, Kubzansky LD. A Systematic Review of the Interplay Between Social Determinants and Environmental Exposures for Early-Life Outcomes. *Curr Environ Health Rep.* 2016 Sep;3 (3):287-301.

28. Pinart M, Benet M, Annesi-Maesano I, von Berg A, Berdel D, Carlsen KC, et al. Comorbidity of eczema, rhinitis, and asthma in IgE-sensitised and non-IgE-sensitised children in MeDALL: a population-based cohort study. *Lancet Respir Med.* 2014 Feb;2 (2):131-40.

29. Ziyab AH, Karmaus W, Zhang H, Holloway JW, Steck SE, Ewart S, et al. Allergic sensitization and filaggrin variants predispose to the comorbidity of eczema, asthma, and rhinitis: results from the Isle of Wight birth cohort. *Clin Exp Allergy*. 2014 Sep;44 (9):1170-8.

30. Yang L, Fu J, Zhou Y. Research Progress in Atopic March. *Front Immunol*. 2020 Aug;11:1-11.

31. Paller AS, Spergel JM, Mina-Osorio P, Irvine AD. The atopic march and atopic multimorbidity: Many trajectories, many pathways. *J Allergy Clin Immunol*. 2019 Jan;143 (1):46-55.

32. Halken S. Prevention of allergic disease in childhood: clinical and epidemiological aspects of primary and secondary allergy prevention. *Pediatr Allergy Immunol.* 2004 Jun;15 Suppl 16:4-5, 9-32.

33. Pinart M, Albang R, Maier D, Duran-Tauleria E, Mena G, Gimeno-Santos E, et al. Systematic Review on the Definition of Allergic Diseases in Children: The MeDALL Study. *Int Arch Allergy Immunol.* 2015 168 (2):110-21.

34. Custovic A, Custovic D, Kljaić Bukvić B, Fontanella S, Haider S. Atopic phenotypes and their implication in the atopic march. *Expert Rev Clin Immunol*. 2020 Sep;16 (9):873-881.

35. McLean WH. The allergy gene: how a mutation in a skin protein revealed a link between eczema and asthma. *F1000 Med Rep.* 2011 Jan 14;3 (2):1-6.

36. Hsu CK, Akiyama M, Shimizu H. Filaggrin: an emerging star in atopic march. *J Formos Med Assoc.* 2008 Jun;107 (6):429-31.

37. Bieber T, D'Erme AM, Akdis CA, Traidl-Hoffmann C, Lauener R, Schappi G, et al. Clinical phenotypes and endophenotypes of atopic dermatitis: Where are we, and where should we go? *J Allergy Clin Immunol*. 2017 Apr;139 (4):58-64.

38. Cosme-Blanco W, Arroyo-Flores E, Ale H. Food Allergies. *Pediatr Rev.* 2020 Aug;41 (8):403-415.

39. Sicherer SH, Warren CM, Dant C, Gupta RS, Nadeau KC. Food Allergy from Infancy Through Adulthood. *J Allergy Clin Immunol Pract.* 2020 Jun;8 (6):1854-1864.

40. Bousquet J, Anto JM, Bachert C, Baiardini I, Bosnic-Anticevich S, Walter Canonica G, et al. Allergic rhinitis. *Nat Rev Dis Primers*. 2020 Dec 3;6 (1):95.

41. Schuler Iv CF, Montejo JM. Allergic Rhinitis in Children and Adolescents. *Immunol Allergy Clin North Am.* 2021 Nov;41 (4):613-625.

42. Tham EH, Leung DY. Mechanisms by Which Atopic Dermatitis Predisposes to Food Allergy and the Atopic March. *Allergy Asthma Immunol Res.* 2019 Jan;11 (1):4-15.

43. McGeachie MJ, Yates KP, Zhou X, Guo F, Sternberg AL, Van Natta ML, et al. Patterns of Growth and Decline in Lung Function in Persistent Childhood Asthma. *N Engl J Med.* 2016 May 12;374 (19):1842-1852.

44. Melen E, Guerra S. Recent advances in understanding lung function development. *F1000Res*. 2017 6 (1):1-11.

45. Schultz ES, Litonjua AA, Melen E. Effects of Long-Term Exposure to Traffic-Related Air Pollution on Lung Function in Children. *Curr Allergy Asthma Rep.* 2017 Jun;17 (6):1-13.

46. Stocks J, Hislop A, Sonnappa S. Early lung development: lifelong effect on respiratory health and disease. *Lancet Respir Med.* 2013 Nov;1 (9):728-42.

47. Quanjer PH, Stocks J, Polgar G, Wise M, Karlberg J, Borsboom G. Compilation of reference values for lung function measurements in children. *Eur Respir J Suppl*. 1989 Mar;4 (1):184-261.

48. Stanojevic S, Wade A, Cole TJ, Lum S, Custovic A, Silverman M, et al. Spirometry centile charts for young Caucasian children: the Asthma UK Collaborative Initiative. *Am J Respir Crit Care Med.* 2009 Sep 15;180 (6):547-52.

49. Moeller A, Carlsen KH, Sly PD, Baraldi E, Piacentini G, Pavord I, et al. Monitoring asthma in childhood: lung function, bronchial responsiveness and inflammation. *Eur Respir Rev.* 2015 Jun;24 (136):204-15. 50. Lesnick BL, Davis SD. Infant pulmonary function testing: overview of technology and practical considerations--new current procedural terminology codes effective 2010. *Chest.* 2011 May;139 (5):1197-1202.

51. Paparella SF. Chloral Hydrate: Safety Risks Still Worth Mentioning. *J Emerg Nurs*. 2018 Jan;44 (1):81-83.

52. Aldrete JA, Itkin IH. Effects of chloral hydrate on the respiration of nonasthmatic and asthmatic patients. *J Allergy*. 1969 Jun;43 (6):342-8.

53. Lodrup Carlsen KC. Tidal breathing at all ages. *Monaldi Arch Chest Dis*. 2000 Oct;55 (5):427-34.

54. van der Ent CK, Brackel HJ, van der Laag J, Bogaard JM. Tidal breathing analysis as a measure of airway obstruction in children three years of age and older. *Am J Respir Crit Care Med.* 1996 Apr;153 (4 Pt 1):1253-8.

55. Seddon PC, Davis GM, Coates AL. Do tidal expiratory flow patterns reflect lung mechanics in infants? *Am J Respir Crit Care Med.* 1996 Apr;153 (4 Pt 1):1248-52.

56. Pijnenburg MW, Baraldi E, Brand PL, Carlsen KH, Eber E, Frischer T, et al. Monitoring asthma in children. *Eur Respir J.* 2015 Apr;45 (4):906-25.

57. Lodrup Carlsen KC, Magnus P, Carlsen KH. Lung function by tidal breathing in awake healthy newborn infants. *Eur Respir J*. 1994 Sep;7 (9):1660-8.

58. Schultz ES, Melen E. Lung development in children: role of genetic and environmental factors. *Acta Paediatr*. 2013 Jul;102 (7):672-4.

59. Haland G, Carlsen KC, Sandvik L, Devulapalli CS, Munthe-Kaas MC, Pettersen M, et al. Reduced lung function at birth and the risk of asthma at 10 years of age. *N Engl J Med.* 2006 Oct 19;355 (16):1682-9.

60. Carlsen KH, Carlson KC, Halvorsen R. Tidal flow volume loops and inflammatory indicators in small children. *Int Arch Allergy Immunol*. 1995 May-Jun;107 (1-3):163-5.

61. Carlsen KH, Lodrup Carlsen KC. Tidal breathing analysis and response to salbutamol in awake young children with and without asthma. *Eur Respir J.* 1994 Dec;7 (12):2154-9.

62. Lodrup Carlsen KC, Mowinckel P, Hovland V, Haland G, Riiser A, Carlsen KH. Lung function trajectories from birth through puberty reflect asthma phenotypes with allergic comorbidity. *J Allergy Clin Immunol*. 2014 Oct;134 (4):917-923.

63. Haland G, Carlsen KH, Devulapalli CS, Pettersen M, Mowinckel P, Lodrup Carlsen KC. Lung function development in the first 2 yr of life is independent of allergic diseases by 2 yr. *Pediatr Allergy Immunol.* 2007 Sep;18 (6):528-34.

64. Martinez FD, Morgan WJ, Wright AL, Holberg C, Taussig LM. Initial airway function is a risk factor for recurrent wheezing respiratory illnesses during the first three years of life. Group Health Medical Associates. *Am Rev Respir Dis.* 1991 Feb;143 (2):312-6.

65. Young S, Arnott J, Le Souef PN, Landau LI. Flow limitation during tidal expiration in symptom-free infants and the subsequent development of asthma. *J Pediatr*. 1994 May;124 (5):681-8.

66. Beydon N, Davis SD, Lombardi E, Allen JL, Arets HG, Aurora P, et al. An official American Thoracic Society/European Respiratory Society statement: pulmonary

function testing in preschool children. *Am J Respir Crit Care Med.* 2007 Jun 15;175 (12):1304-45.

67. Stocks J, Dezateux CA, Jackson EA, Hoo AF, Costeloe KL, Wade AM. Analysis of tidal breathing parameters in infancy: how variable is TPTEF:TE? *Am J Respir Crit Care Med.* 1994 Nov;150 (5):1347-54.

68. Anık A, Uysal P. Impaired Lung Functions Using Tidal Breath Analysis in High-risk Infants with Recurrent Wheezing. *TIP*. 2020 1 (2):49-54.

69. Milner AD. Lung function testing in infancy. *Arch Dis Child*. 1990 May;65 (5):548-52.

70. Lødrup KC, Mowinckel P, Carlsen KH. Lung function measurements in awake compared to sleeping newborn infants. *Pediatr Pulmonol*. 1992 Feb;12 (2):99-104.

71. Marenholz I, Esparza-Gordillo J, Lee YA. The genetics of the skin barrier in eczema and other allergic disorders. *Curr Opin Allergy Clin Immunol*. 2015 Oct;15 (5):426-34.

72. van den Wijngaart LS, Roukema J, Merkus PJ. Respiratory disease and respiratory physiology: putting lung function into perspective: paediatric asthma. *Respirology*. 2015 Apr;20 (3):379-88.

73. Kubo A, Nagao K, Amagai M. Epidermal barrier dysfunction and cutaneous sensitization in atopic diseases. *J Clin Invest*. 2012 Feb;122 (2):440-7.

74. Akdis CA. Does the epithelial barrier hypothesis explain the increase in allergy, autoimmunity and other chronic conditions? *Nat Rev Immunol.* 2021 Nov;21 (11):739-751.

75. Reynolds LA, Finlay BB. Early life factors that affect allergy development. *Nat Rev Immunol.* 2017 Aug;17 (8):518-528.

76. Kusari A, Han AM, Schairer D, Eichenfield LF. Atopic Dermatitis: New Developments. *Dermatol Clin.* 2019 Jan;37 (1):11-20.

77. Weidinger S, O'Sullivan M, Illig T, Baurecht H, Depner M, Rodriguez E, et al. Filaggrin mutations, atopic eczema, hay fever, and asthma in children. *J Allergy Clin Immunol.* 2008 May;121 (5):1203-1209.

78. Kim BE, Leung DYM. Significance of Skin Barrier Dysfunction in Atopic Dermatitis. *Allergy Asthma Immunol Res.* 2018 May;10 (3):207-215.

79. Cubero JL, Isidoro-Garcia M, Segura N, Benito Pescador D, Sanz C, Lorente F, et al. Filaggrin gene mutations and new SNPs in asthmatic patients: a cross-sectional study in a Spanish population. *Allergy Asthma Clin Immunol.* 2016 12 (31):1-12.

80. Elias PM. Stratum corneum defensive functions: an integrated view. *J Invest Dermatol.* 2005 Aug;125 (2):183-200.

81. Knight DA, Holgate ST. The airway epithelium: structural and functional properties in health and disease. *Respirology*. 2003 Dec;8 (4):432-46.

82. Ahrens B, Posa D. Prevention of allergies in childhood - where are we now? *Allergol Select.* 2017 1 (2):200-213.

83. Suzuki Y, Kodama M, Asano K. Skin barrier-related molecules and pathophysiology of asthma. *Allergol Int.* 2011 Mar;60 (1):11-5.

84. Mims JW. Asthma: definitions and pathophysiology. *Int Forum Allergy Rhinol*. 2015 Sep;5 (1):2-6.

85. Noutsios GT, Floros J. Childhood asthma: causes, risks, and protective factors; a role of innate immunity. *Swiss Med Wkly*. 2014 Dec;144 (1):1-14.

86. Lee YJ, Fujisawa T, Kim CK. Biomarkers for Recurrent Wheezing and Asthma in Preschool Children. *Allergy Asthma Immunol Res.* 2019 Jan;11 (1):16-28.

87. Fitzpatrick AM. Severe Asthma in Children: Lessons Learned and Future Directions. *J Allergy Clin Immunol Pract.* 2016 Jan-Feb;4 (1):11-9.

88. Wechsler ME, Munitz A, Ackerman SJ, Drake MG, Jackson DJ, Wardlaw AJ, et al. Eosinophils in Health and Disease: A State-of-the-Art Review. *Mayo Clin Proc.* 2021 Oct;96 (10):2694-2707.

89. Komlósi ZI, van de Veen W, Kovács N, Szűcs G, Sokolowska M, O'Mahony L, et al. Cellular and molecular mechanisms of allergic asthma. *Mol Aspects Med.* 2022 Jun;85 (1):1-30.

90. Kim CK. Eosinophil-derived neurotoxin: a novel biomarker for diagnosis and monitoring of asthma. *Korean J Pediatr*. 2013 Jan;56 (1):8-12. In File.

91. Wolthers OD. Eosinophil granule proteins in the assessment of airway inflammation in pediatric bronchial asthma. *Pediatr Allergy Immunol.* 2003 Aug;14 (4):248-54.

92. Stone KD, Prussin C, Metcalfe DD. IgE, mast cells, basophils, and eosinophils. *J Allergy Clin Immunol.* 2010 Feb;125 (2):73-80.

93. Malinovschi A, Rydell N, Fujisawa T, Borres MP, Kim CK. Clinical Potential of Eosinophil-Derived Neurotoxin in Asthma Management. *J Allergy Clin Immunol Pract.* 2023 Mar;11 (3):750-761.

94. Koh GC, Shek LP, Goh DY, Van Bever H, Koh DS. Eosinophil cationic protein: is it useful in asthma? A systematic review. *Respir Med.* 2007 Apr;101 (4):696-705.

95. Koller DY, Wojnarowski C, Herkner KR, Weinländer G, Raderer M, Eichler I, et al. High levels of eosinophil cationic protein in wheezing infants predict the development of asthma. *J Allergy Clin Immunol.* 1997 Jun;99 (6):752-6.

96. Nja F, Røksund OD, Carlsen KH. Eosinophil cationic protein (ECP) in school children living in a mountainous area of Norway: a population-based study of ECP as a tool for diagnosing asthma in children with reference values. *Allergy*. 2001 Feb;56 (2):138-44.

97. Joseph-Bowen J, de Klerk N, Holt PG, Sly PD. Relationship of asthma, atopy, and bronchial responsiveness to serum eosinophil cationic proteins in early childhood. *J Allergy Clin Immunol.* 2004 Nov;114 (5):1040-5.

98. Kristjánsson S, Shimizu T, Strannegård IL, Wennergren G. Eosinophil cationic protein, myeloperoxidase and tryptase in children with asthma and atopic dermatitis. *Pediatr Allergy Immunol.* 1994 Nov;5 (4):223-9.

99. Rydell N, Nagao M, Movérare R, Ekoff H, Sjölander A, Borres MP, et al. Serum Eosinophilic Cationic Protein Is a Reliable Biomarker for Childhood Asthma. *Int Arch Allergy Immunol.* 2022 183 (7):744-752. 100. Prehn A, Seger RA, Faber J, Torresani T, Molinari L, Gerber A, et al. The relationship of serum-eosinophil cationic protein and eosinophil count to disease activity in children with bronchial asthma. *Pediatr Allergy Immunol*. 1998 Nov;9 (4):197-203.

101. Remes S, Korppi M, Remes K, Savolainen K, Mononen I, Pekkanen J. Serum eosinophil cationic protein (ECP) and eosinophil protein X (EPX) in childhood asthma: the influence of atopy. *Pediatr Pulmonol.* 1998 Mar;25 (3):167-74.

102. Wu KG, Li TH, Chen CJ, Cheng HI, Wang TY. Correlations of serum Interleukin-16, total IgE, eosinophil cationic protein and total eosinophil counts with disease activity in children with atopic dermatitis. *Int J Immunopathol Pharmacol*. 2011 Jan-Mar;24 (1):15-23.

103. Lødrup Carlsen KC, Halvorsen R, Ahlstedt S, Carlsen KH. Eosinophil cationic protein and tidal flow volume loops in children 0-2 years of age. *Eur Respir J.* 1995 Jul;8 (7):1148-54.

104. Rutten B, Young S, Rhedin M, Olsson M, Kurian N, Syed F, et al. Eosinophilderived neurotoxin: A biologically and analytically attractive asthma biomarker. *PLoS One*. 2021 Feb;16 (2):e0246627.

105. Hoekstra MO, Hovenga H, Gerritsen J, Kauffman HF. Eosinophils and eosinophil-derived proteins in children with moderate asthma. *Eur Respir J*. 1996 Nov;9 (11):2231-5.

106. Kim CK, Callaway Z, Fletcher R, Koh YY. Eosinophil-derived neurotoxin in childhood asthma: correlation with disease severity. *J Asthma*. 2010 Jun;47 (5):568-73.

107. Kim KW, Lee KE, Kim ES, Song TW, Sohn MH, Kim KE. Serum eosinophilderived neurotoxin (EDN) in diagnosis and evaluation of severity and bronchial hyperresponsiveness in childhood asthma. *Lung.* 2007 Mar-Apr;185 (2):97-103.

108. Koller DY, Halmerbauer G, Frischer T, Roithner B. Assessment of eosinophil granule proteins in various body fluids: is there a relation to clinical variables in childhood asthma? *Clin Exp Allergy*. 1999 Jun;29 (6):786-93.

109. Rydell N, Nagao M, Ekoff H, Sjölander A, Borres M, Fujisawa T. Development of an automated ImmunoCAP research assay for eosinophil derived neurotoxin and its use in asthma diagnosis in children. *Pract Lab Med.* 2019 Nov;17:e00138.

110. Kim CK, Kang DY, Callaway Z, Kim KS, Kwon EM, Yamaide F, et al. Increase in eosinophil-derived neurotoxin level in school children with allergic disease. *Asia Pac Allergy*. 2022 Jul;12 (3):1-9.

111. Niggemann B, Kleinau I, Schmitt M, Wahn U. Twenty-four-hour time course of eosinophil granule proteins ECP and EPX during bronchial allergen challenges in serum of asthmatic children. *Allergy*. 1994 Feb;49 (2):74-80.

112. Rao R, Frederick JM, Enander I, Gregson RK, Warner JA, Warner JO. Airway function correlates with circulating eosinophil, but not mast cell, markers of inflammation in childhood asthma. *Clin Exp Allergy*. 1996 Jul;26 (7):789-93.

113. Zimmerman B, Enander I, Zimmerman R, Ahlstedt S. Asthma in children less than 5 years of age: eosinophils and serum levels of the eosinophil proteins ECP and EPX in relation to atopy and symptoms. *Clin Exp Allergy*. 1994 Feb;24 (2):149-55.

114. Kim H, Kwon GE, Kim YH, Callaway Z, Han YS, Seo JJ, et al. Comparison of serum eosinophil-derived neurotoxin levels between wheezing and non-wheezing groups in children with respiratory tract infection. *J Asthma*. 2020 Nov;57 (11):1211-15.

115. Kim CK, Choi J, Kim HB, Callaway Z, Shin BM, Kim JT, et al. A randomized intervention of montelukast for post-bronchiolitis: effect on eosinophil degranulation. *J Pediatr.* 2010 May;156 (5):749-54.

116. Almqvist C, Worm M, Leynaert B. Impact of gender on asthma in childhood and adolescence: a GA2LEN review. *Allergy*. 2008 Jan;63 (1):47-57.

117. Raghavan D, Jain R. Increasing awareness of sex differences in airway diseases. *Respirology*. 2016 Apr;21 (3):449-59.

118. Chowdhury NU, Guntur VP, Newcomb DC, Wechsler ME. Sex and gender in asthma. *Eur Respir Rev.* 2021 Dec 31;30 (162):1-14.

119. Lødrup Carlsen KC, Rehbinder EM, Skjerven HO, Carlsen MH, Fatnes TA, Fugelli P, et al. Preventing Atopic Dermatitis and ALLergies in Children-the PreventADALL study. *Allergy*. 2018 Oct;73 (10):2063-70.

120. Skjerven HO, Rehbinder EM, Vettukattil R, LeBlanc M, Granum B, Haugen G, et al. Skin emollient and early complementary feeding to prevent infant atopic dermatitis (PreventADALL): a factorial, multicentre, cluster-randomised trial. *Lancet.* 2020 Mar 21;395 (10228):951-961.

121. Skjerven HO, Lie A, Vettukattil R, Rehbinder EM, LeBlanc M, Asarnoj A, et al. Early food intervention and skin emollients to prevent food allergy in young children (PreventADALL): a factorial, multicentre, cluster-randomised trial. *Lancet*. 2022 Jun 25;399 (10344):2398-2411.

122. Bates JH, Schmalisch G, Filbrun D, Stocks J. Tidal breath analysis for infant pulmonary function testing. ERS/ATS Task Force on Standards for Infant Respiratory Function Testing. European Respiratory Society/American Thoracic Society. *Eur Respir J.* 2000 Dec;16 (6):1180-92.

123. Bains KES, Gudmundsdóttir HK, Färdig M, Amnö E, Jonassen CM, Nordlund B, et al. Infant lung function: criteria for selecting tidal flow-volume loops. *ERJ Open Res.* 2022 Oct;8 (4):1-11.

124. Rehbinder EM, Advocaat Endre KM, Lødrup Carlsen KC, Asarnoj A, Stensby Bains KE, Berents TL, et al. Predicting Skin Barrier Dysfunction and Atopic Dermatitis in Early Infancy. *J Allergy Clin Immunol Pract.* 2020 Feb;8 (2):664-673.

125. Brenninkmeijer EE, Schram ME, Leeflang MM, Bos JD, Spuls PI. Diagnostic criteria for atopic dermatitis: a systematic review. *Br J Dermatol*. 2008 Apr;158 (4):754-65.

126. Berents TL, Carlsen KC, Mowinckel P, Skjerven HO, Kvenshagen B, Rolfsjord LB, et al. Skin Barrier Function and Staphylococcus aureus Colonization in Vestibulum Nasi and Fauces in Healthy Infants and Infants with Eczema: A Population-Based Cohort Study. *PLoS One.* 2015 10 (6):e0130145.

127. Hoyer A, Rehbinder EM, Färdig M, Asad S, Lødrup Carlsen KC, Endre KMA, et al. Filaggrin mutations in relation to skin barrier and atopic dermatitis in early infancy. *Br J Dermatol.* 2022 Mar;186 (3):544-552.

128. Lodrup Carlsen KC, Carlsen KH, Nafstad P, Bakketeig L. Perinatal risk factors for recurrent wheeze in early life. *Pediatr Allergy Immunol*. 1999 May;10 (2):89-95.

129. Hevroni A, Goldman A, Blank-Brachfeld M, Abu Ahmad W, Ben-Dov L, Springer C. Use of tidal breathing curves for evaluating expiratory airway obstruction in infants. *Journal of Asthma*. 2018 2018/12/02;55 (12):1331-1337.

130. Morris MJ, Lane DJ. Tidal expiratory flow patterns in airflow obstruction. *Thorax.* 1981 Feb;36 (2):135-42.

131. Wärnberg Gerdin S, Lie A, Asarnoj A, Borres MP, Lødrup Carlsen KC, Färdig M, et al. Impaired skin barrier and allergic sensitization in early infancy. *Allergy*. 2022 May;77 (5):1464-1476.

132. Endre KMA, Landrø L, LeBlanc M, Gjersvik P, Lødrup Carlsen KC, Haugen G, et al. Diagnosing atopic dermatitis in infancy using established diagnostic criteria: a cohort study. *Br J Dermatol.* 2022 Jan;186 (1):50-58.

133. Lødrup Carlsen KC, Roll S, Carlsen KH, Mowinckel P, Wijga AH, Brunekreef B, et al. Does pet ownership in infancy lead to asthma or allergy at school age? Pooled analysis of individual participant data from 11 European birth cohorts. *PLoS One*. 2012 7 (8):e43214.

134. Hajian-Tilaki K. The choice of methods in determining the optimal cut-off value for quantitative diagnostic test evaluation. *Stat Methods Med Res.* 2018 Aug;27 (8):2374-83.

135. World Medical Association (WMA). WMA Declaration of Helsinki – Ethical principles for medical research involving human subjects. Accessed 25 May, 2023. https://www.wma.net/policies-post/wma-declaration-of-helsinki-ethical-principles-for-medical-research-involving-human-subjects/

136. Bains KES, Färdig M, Gudmundsdóttir HK, Almqvist C, Hedlin G, Nordhagen LS, et al. Infant tidal flow-volume parameters and arousal state. *ERJ Open Res.* 2022 Oct;8 (4):1-8.

137. Färdig M, Gudmundsdóttir HK, Hoyer A, Bains KES, Almqvist C, Monceyron Jonassen C, et al. Skin Barrier Function and Infant Tidal Flow-Volume Loops-A Population-Based Observational Study. *Children (Basel)*. 2022 Dec 31;10 (1):1-13.

138. Micallef L, Rodgers P. eulerAPE: drawing area-proportional 3-Venn diagrams using ellipses. *PLoS One*. 2014 9 (7):e101717.

139. Lodrup KC, Mowinckel P, Carlsen KH. Lung function measurements in awake compared to sleeping newborn infants. *Pediatr Pulmonol.* 1992 Feb;12 (2):99-104.

140. Heraghty JL, Hilliard TN, Henderson AJ, Fleming PJ. The physiology of sleep in infants. *Arch Dis Child*. 2008 Nov;93 (11):982-5.

141. Gracia-Tabuenca J, Seppä VP, Jauhiainen M, Paassilta M, Viik J, Karjalainen J. Tidal breathing flow profiles during sleep in wheezing children measured by impedance pneumography. *Respir Physiol Neurobiol*. 2020 Jan;271:1-8.

142. Krieger J. Breathing during sleep in normal subjects. *Clin Chest Med.* 1985 Dec;6 (4):577-94.

143. Francisco CO, Bhatawadekar SA, Babineau J, Reid WD, Yadollahi A. Effects of physical exercise training on nocturnal symptoms in asthma: Systematic review. *PLoS One.* 2018 13 (10):e0204953.

144. Rusconi F, Gagliardi L, Aston H, Silverman M. Changes in respiratory rate affect tidal expiratory flow indices in infants with airway obstruction. *Pediatr Pulmonol*. 1996 Apr;21 (4):236-40.

145. Guerra S, Lombardi E, Stern DA, Sherrill DL, Gilbertson-Dahdal D, Wheatley-Guy CM, et al. Fetal Origins of Asthma: A Longitudinal Study from Birth to Age 36 Years. *Am J Respir Crit Care Med.* 2020 Dec 15;202 (12):1646-1655.

146. Çelik E, Uysal P. Pulmonary function testing with tidal breath analyze technique is useful in predicting persistant small airway damage in infants with acute bronchiolitis. *Pediatr Allergy Immunol.* 2021 Jan;32 (1):60-66.

147. Löffler H, Effendy I. Skin susceptibility of atopic individuals. *Contact Dermatitis*. 1999 May;40 (5):239-42.

148. Hu C, Nijsten T, van Meel ER, Erler NS, Piketty C, de Jong NW, et al. Eczema phenotypes and risk of allergic and respiratory conditions in school age children. *Clin Transl Allergy*. 2020 10 (7):1-8.

149. Håland G, Carlsen KH, Devulapalli CS, Pettersen M, Mowinckel P, Lødrup Carlsen KC. Lung function development in the first 2 yr of life is independent of allergic diseases by 2 yr. *Pediatr Allergy Immunol.* 2007 Sep;18 (6):528-34.

150. Amat F, Soria A, Tallon P, Bourgoin-Heck M, Lambert N, Deschildre A, et al. New insights into the phenotypes of atopic dermatitis linked with allergies and asthma in children: An overview. *Clin Exp Allergy*. 2018 Aug;48 (8):919-934.

151. Basu K, Inglis SK, Bremner SA, Ramsay R, Abd A, Rabe H, et al. Filaggrin gene defects are associated with eczema, wheeze, and nasal disease during infancy: Prospective study. *J Allergy Clin Immunol.* 2020 Sep;146 (3):681-682.

152. Müller S, Marenholz I, Lee YA, Sengler C, Zitnik SE, Griffioen RW, et al. Association of Filaggrin loss-of-function-mutations with atopic dermatitis and asthma in the Early Treatment of the Atopic Child (ETAC) population. *Pediatr Allergy Immunol.* 2009 Jun;20 (4):358-61.

153. Ponińska J, Samoliński B, Tomaszewska A, Raciborski F, Samel-Kowalik P, Walkiewicz A, et al. Filaggrin gene defects are independent risk factors for atopic asthma in a Polish population: a study in ECAP cohort. *PLoS One*. 2011 Feb 18;6 (2):e16933.

154. Sugiura S, Hiramitsu Y, Futamura M, Kamioka N, Yamaguchi C, Umemura H, et al. Clinical risk factors at 3 months of age for the development of bronchial asthma at 36 months of age. *Pediatr Int.* 2023 Mar 17:e15530.

155. Lee E, Lee SH, Kwon JW, Kim YH, Yoon J, Cho HJ, et al. Persistent asthma phenotype related with late-onset, high atopy, and low socioeconomic status in school-aged Korean children. *BMC Pulm Med.* 2017 Feb 23;17 (1):1-11.

156. von Kobyletzki LB, Bornehag CG, Hasselgren M, Larsson M, Lindström CB, Svensson Å. Eczema in early childhood is strongly associated with the development of asthma and rhinitis in a prospective cohort. *BMC Dermatol.* 2012 Jul 27;12 (11):1-9.

157. Just J, Saint Pierre P, Amat F, Gouvis-Echraghi R, Lambert-Guillemot N, Guiddir T, et al. What lessons can be learned about asthma phenotypes in children from cohort studies? *Pediatr Allergy Immunol.* 2015 Jun;26 (4):300-5.

158. Stocks J, Henschen M, Hoo AF, Costeloe K, Dezateux C. Influence of ethnicity and gender on airway function in preterm infants. *Am J Respir Crit Care Med.* 1997 Dec;156 (6):1855-62.

159. Dvornyk V, Ponomarenko I, Belyaeva T, Reshetnikov E, Churnosov M. Filaggrin gene polymorphisms are associated with atopic dermatitis in women but not in men in the Caucasian population of Central Russia. *PLoS One*. 2021 16 (12):e0261026.

160. Lowe AJ, Carlin JB, Bennett CM, Hosking CS, Abramson MJ, Hill DJ, et al. Do boys do the atopic march while girls dawdle? *J Allergy Clin Immunol*. 2008 May;121 (5):1190-5.

161. Granger V, Zerimech F, Arab J, Siroux V, de Nadai P, Tsicopoulos A, et al. Blood eosinophil cationic protein and eosinophil-derived neurotoxin are associated with different asthma expression and evolution in adults. *Thorax*. 2022 Jun;77 (6):552-62.

162. Rydell N, Nagao M, Ekoff H, Sjölander A, Borres M, Fujisawa T. Development of an automated ImmunoCAP research assay for eosinophil derived neurotoxin and its use in asthma diagnosis in children. *Pract Lab Med.* 2019 Nov;17 (1):e00138.

163. Chakraborty S, Hammar KS, Filiou AE, Holmdahl I, Hoyer A, Ekoff H, et al. Longitudinal eosinophil-derived neurotoxin measurements and asthma development in preschool wheezers. *Clin Exp Allergy*. 2022 Jul;52 (11):1338-42.

164. Amaral R, Jacinto T, Malinovschi A, Janson C, Price D, Fonseca JA, et al. The influence of individual characteristics and non-respiratory diseases on blood eosinophil count. *Clin Transl Allergy*. 2021 Jun;11 (4):e12036.

165. Mogensen I, Alving K, Dahlen SE, James A, Forsberg B, Ono J, et al. Fixed airflow obstruction relates to eosinophil activation in asthmatics. *Clin Exp Allergy*. 2019 Feb;49 (2):155-162.

166. Case LD, Ambrosius WT. Power and sample size. *Methods Mol Biol.* 2007 404:377-408.

167. American Thoracic Society/European Respiratory Society. Respiratory mechanics in infants: physiologic evaluation in health and disease. *Am Rev Respir Dis.* 1993 Feb;147 (2):474-96.

168. Lødrup Carlsen KC, Rehbinder EM, Skjerven HO, Carlsen MH, Fatnes TA, Fugelli P, et al. Preventing Atopic Dermatitis and ALLergies in Children-the PreventADALL study. *Allergy*. 2018 Oct;73 (10):2063-2070.

169. Leed JE, Chinn LK, Lockman JJ. Reaching to the Self: The Development of Infants' Ability to Localize Targets on the Body. *Psychol Sci.* 2019 Jul;30 (7):1063-1073.

170. Rutten B, Young S, Aldridge S, Davies R, Fidock M, Newbold P, et al. The commutability of enzyme linked immunosorbent assays for the quantification of serum eosinophil-derived neurotoxin (EDN). *J Immunol Methods*. 2022 Jan;500 (1):1-5.

171. Vrbova M, Dorociakova P, Vyskovsky R, Palkovicova Murinova L, Ciznar P, Rausova K, et al. Dynamics of allergy development during the first 5 years of life. *Eur J Pediatr.* 2018 Sep;177 (9):1317-1325.

172. Lødrup Carlsen KC, Halvorsen R, Carlsen KH. Serum inflammatory markers and effects of age and tobacco smoke exposure in young non-asthmatic children. *Acta Paediatr.* 1998 May;87 (5):559-64.

173. Knipping K, Knippels LM, Dupont C, Garssen J. Serum biomarkers for allergy in children. *Pediatr Allergy Immunol.* 2017 Mar;28 (2):114-23.

174. Licari A, Castagnoli R, Brambilla I, Marseglia A, Tosca MA, Marseglia GL, et al. Asthma Endotyping and Biomarkers in Childhood Asthma. *Pediatr Allergy Immunol Pulmonol.* 2018 Jun 1;31 (2):44-55.

175. Rodriguez Del Rio P, Liu AH, Borres MP, Södergren E, Iachetti F, Casale TB. Asthma and Allergy: Unravelling a Tangled Relationship with a Focus on New Biomarkers and Treatment. *Int J Mol Sci.* 2022 Mar;23 (7):3881.

176. Anık A, Öztürk S, Erge D, Akcan AB, Türkmen MK, Uysal P. Tidal breath in healthy term newborns: An analysis from the 2nd to the 30th days of life. *Pediatr Pulmonol*. 2021 Jan;56 (1):274-282.

177. Rusconi F, Castagneto M, Gagliardi L, Leo G, Pellegatta A, Porta N, et al. Reference values for respiratory rate in the first 3 years of life. *Pediatrics*. 1994 Sep;94 (3):350-5.